DUFF&PHELPS

First Report to Court of the Receiver of blutip Power Technologies Ltd.

March 9, 2012

Contents

		Page
Introd	luction	1
1.1	Purposes of this Report	2
1.2	Currency	2
Back	ground	3
2.1		
The F	Receivership Order	4
3.1	Priority of Charges	4
CCM		4
4.1	Security Opinion	8
The S	Stalking Horse Offer	8
Sale	Process and Bidding Procedures	9
6.1	Sale Process Recommendation	12
Over	view of the Receiver's Activities	13
Conc	lusion and Recommendation	14
ndices		Tab
ership	Order and Endorsement dated February 28, 2012	A
ns Co 8, 201	mpany dated March 2, 2012, from counsel to Sparta Capital Ltd. 2 (without attachments) and from The Cell, Inc. dated January 17,	dated , 2012
	,	
	1.1 1.2 Backy 2.1 The F 3.1 CCM 4.1 The S Sale 6.1 Overv Conc	1.2 Currency Background 2.1 Intellectual Property. The Receivership Order. 3.1 Priority of Charges. CCM. 4.1 Security Opinion. The Stalking Horse Offer. Sale Process and Bidding Procedures. 6.1 Sale Process Recommendation. Overview of the Receiver's Activities. Conclusion and Recommendation.



Court File No.: CV-12-9622-00CL

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

BETWEEN:

CCM Master Qualified Fund, Ltd.

Applicant

-and-

blutip Power Technologies Ltd.

Respondent

IN THE MATTER OF SECTION 243(1) OF THE BANKRUPTCY AND INSOLVENCY ACT, R.S.C. 1985, c. B-3, AS AMENDED

FIRST REPORT OF THE RECEIVER

March 9, 2012

1.0 Introduction

Pursuant to an order of the Ontario Superior Court of Justice (the "Court") made on February 28, 2012 (the "Receivership Order"), Duff & Phelps Canada Restructuring Inc. ("D&P") was appointed receiver ("Receiver") of the properties, assets and undertakings (collectively, the "Assets") of blutip Power Technologies Ltd. (the "Company") pursuant to Section 243(1) of the *Bankruptcy and Insolvency Act*, R.S.C. 1985, c. B-3, as amended ("BIA"). A copy of the Receivership Order and Endorsement dated February 28, 2012 is attached as Appendix "A".

This report ("Report") is filed by D&P in its capacity as Receiver.

The primary purpose of these receivership proceedings is to allow for the Company's business and assets to be marketed for sale pursuant to a Court-supervised sale process ("Sale Process").

Duff & Phelps | Page 1 of 14

1.1 Purposes of this Report

The purposes of this Report are to:

- Provide background information concerning the Company and these receivership proceedings;
- Summarize the Company's indebtedness to CCM Master Qualified Fund, Ltd. ("CCM"), the Company's sole secured creditor, including the results of an opinion dated March 9, 2012 ("Opinion") rendered by Blake Cassels & Graydon LLP ("Blakes"), the Receiver's legal counsel, regarding the validity and enforceability of CCM's security;
- Summarize the terms of an offer submitted by CCM to the Receiver on March 3, 2012 to purchase substantially all of the Company's business and Assets (the "Stalking Horse Offer"), which, subject to the approval of this Court, would act as a "stalking horse" in the Sale Process;
- Summarize the proposed Sale Process pursuant to which the Company's business and Assets would be marketed for sale, including the bidding procedures to be used in connection with the Sale Process (the "Bidding Procedures");
- Provide an overview of the Receiver's activities since the commencement of these proceedings; and
- Recommend that the Court issue an order:
 - a) Ordering that the Receiver's Charge and the Receiver's Borrowings Charge (each as defined in the Receivership Order) have standard priority of such charges;
 - b) Approving the Stalking Horse Offer;
 - c) Approving the Sale Process, including the Bidding Procedures, and authorizing and directing the Receiver to conduct the Sale Process on the basis detailed herein; and
 - d) Approving the Receiver's activities, as described in this Report.

1.2 Currency

All currency references in this Report are to Canadian dollars, unless otherwise noted.

Duff & Phelps | Page 2 of 14

2.0 Background

The Company is a publicly listed ¹ technology company (TSX-V: BPR) based in Mississauga, Ontario. The Company engages in the research, development and sale of hydrogen generating systems and combustion controls. Its products include hydrogen generating systems that inject small amounts of hydrogen gas, on demand, into the combustion chamber of a vehicle's engine. These products enrich the fuel-air charge of an internal combustion engine with hydrogen produced through electrolysis and improve combustion of the fuel-air mixture. The result is improved fuel economy in large diesel engines, reducing the greenhouse gas output and carbon footprint of such engines, and cost savings for engine operators.

The Company presently employs ten individuals. The Company's workforce is not unionized and the Company does not maintain any pension plans.

Further information about the Company and its background was provided in the affidavit of James Schuler sworn February 27, 2012 (the "Affidavit") included in the receivership application materials. A copy of the Affidavit (without exhibits) is attached as Appendix "B".

2.1 Intellectual Property

The Company is in its development phase. It has invested in the research and development of hydrogen generating systems and control technology resulting in proprietary, patented and patent-pending technology. A list of certain patents registered in the Company's name is attached as Schedule "C" to the Stalking Horse Offer. Additional information concerning the Company's patents, trademarks and other intellectual property will be made available to prospective purchasers in the data room (discussed in Section 6 below).

The Receiver is aware of certain proprietary and damage claims made by third parties against the Company, including with respect to certain of its intellectual property. In this regard, a letter dated February 27, 2012 from ICE Fuel Technologies Ltd. to the Company, a letter dated March 2, 2012 from Aurora Power Solutions Company to the Receiver, a letter dated March 8, 2012 from counsel to Sparta Capital Ltd. to the Receiver (without attachments) and an email dated January 17, 2012 from The Cell, Inc. to the Company (without attachments) are attached as Appendix "C".

Duff & Phelps | Page 3 of 14

¹ Effective on or around February 8, 2012, the TSX Venture Exchange suspended the Company from trading for failing to maintain exchange listing requirements.

If the Sale Process detailed herein is approved by the Court, the Receiver will provide further information to potential bidders with respect to these and other claims made by third parties relating to intellectual property. Any proposed sale by the Receiver would be on an "as is, where is" basis and would seek to convey the Receiver's and the Company's right, title and interest, if any, in the assets subject to the sale.

3.0 The Receivership Order

3.1 Priority of Charges

The hearing to obtain the Receivership Order was brought by CCM on an urgent, ex parte basis for the reasons detailed in the Affidavit. Accordingly, priority over existing perfected security interests and statutory encumbrances was not sought for the Receiver's Charge or the Receiver's Borrowings Charge.

This Report and the related motion materials are being served on:

- (i) all parties with registered security interests pursuant to the *Personal Property Security Act* (Ontario) ("PPSA") and the *Personal Property Security Act* (Alberta) (the Company is headquartered in Ontario and was originally incorporated in Alberta);
- (ii) all parties who have commenced legal proceedings against the Company or are co-defendants with the Company in such proceedings (as revealed through litigation searches);
- (iii) all parties who have asserted claims against the Company with respect to intellectual property;
- (iv) the Company's landlord; and
- (v) standard government agencies, including Canada Revenue Agency and the Ministry of Finance for the Province of Ontario.

As interested parties are being served, the Receiver respectfully requests that this Court order that the Receiver's Charge and the Receiver's Borrowings Charge have the standard priority of such charges.

4.0 CCM

Based on the Company's books and records, CCM is the Company's sole secured creditor. ² According to the Company's books and records, the Company's

Duff & Phelps | Page 4 of 14

_

² The Receiver has been advised that Blutip Financial Corporation has asserted a claim, based on disputed facts, that it has priority over CCM's security based on equitable principles. The claim has not been adjudicated.

indebtedness to CCM presently totals approximately \$3.5 million, comprised of the following:

- A convertible senior secured promissory note dated October 21, 2011 in the principal amount of \$2.6 million (the "October Note");
- A convertible senior secured promissory note dated December 29, 2011 in the principal amount of \$800,000 (the "December Note"); and
- \$65,000 advanced by CCM to the Receiver on February 29, 2012 pursuant to a term sheet in the principal amount of \$400,000 dated February 27, 2012 between the Receiver and CCM (the "Term Sheet"). On February 29, 2012, the Receiver issued a Receiver's Certificate to CCM evidencing this advance.

The October Note and the December Note are jointly referred to herein as the "Notes". The term of the October Note is two years and the term of the December Note is 22 months. Both Notes bear interest at 15% per annum. As detailed in the Affidavit, the proceeds of the Notes were used to, *inter alia*, "pre-pay" interest and fees and a portion of the October Note proceeds was used to discharge certain unsecured obligations owing by the Company to CCM. Based on the Company's books and records, the following table reflects the payments funded from the proceeds of the Notes:

Description	October Note	December Note	Total
Principal and interest on existing secured notes in favour of third parties not related to CCM ⁵	1,020,333	-	1,020,333
Repayments to CCM of existing unsecured notes and advances ⁶	157,021	_	157,021
Prepaid interest on the Notes	780,000	220,932	1,000,932
Commitment fee	-	45,000	45,000
Legal fees	20,000	3,000	23,000
General working capital purposes	622,646	531,068	1,153,714
Total	2,600,000	800,000	3,400,000

Duff & Phelps | Page 5 of 14

³ By agreement between the Receiver and CCM, the Term Sheet maturity date is being extended from April 30, 2012 to May 15, 2012 (the date on which all bids pursuant to the Sale Process become irrevocable).

⁴ The Notes also provide for the rate of interest to be increased by 2% per annum during the continuation of an Event of Default (as defined therein). CCM has agreed, on a without prejudice basis, to waive such default interest.

⁵ Represents repayment of principal and interest owing on four \$250,000 convertible, secured notes issued on June 30, 2011 by the Company in favour of John S. Chambers, Steven Collicut, Kevin Bennett and Black Swan Advisors Inc.

⁶ Represents repayment of principal and interest to CCM pursuant to unsecured promissory notes dated September 30, 2011 and October 14, 2011 in the amounts of \$76,000 and \$30,000, respectively, and and unsecured advance of \$50,000 made by CCM to the Company on October 24, 2011.

The Receiver notes the following in respect of certain payments reflected in the table above:

The interest on the Notes was prepaid. Accordingly, the amount which continues to be outstanding under the Notes is \$3.4 million as no further interest is accruing on the Notes. If the Stalking Horse Offer is the "Successful Bid" in accordance with the Bidding Procedures, it is anticipated that the transaction contemplated in the Stalking Horse Offer will close on May 3, 2012. Accordingly, the Notes would effectively be repaid on May 3, 2012. There would be no refund of any prepaid interest if the Notes are paid prior to maturity.

The Receiver is cognizant that if the term of the Notes is substantially shortened, the interest paid on the Notes may be in breach of Section 347 of the *Criminal Code* (Canada), which provides that the effective rate of interest shall not exceed 60% per annum. The Receiver has considered what the effective rate of interest under the Notes would be, on an annual basis, based on this shortened term (i.e. with the Notes being effectively repaid on May 3, 2012). Additional analysis on the interpretation of Section 347 of the *Criminal Code* (Canada) will be set out in the factum to be filed by Blakes, on the Receiver's behalf.

If the Stalking Horse Offer were to close on May 3, 2012, CCM would have collected: (i) in respect of the October Note, \$780,000 in interest for a loan that was outstanding between October 21, 2011 and May 3, 2012. In addition, \$20,000 of the proceeds from the October Note was used to pay CCM's legal fees. If the legal fees were characterized as interest, the effective annual interest rate on the October Note, if repaid in full on May 3, 2012, would be 57.6%; and (ii) in respect of the December Note, \$220,932 in interest on a loan that was outstanding between December 29, 2011 and May 3, 2012, resulting in an effective annual rate of interest of 80.0%. In addition, the December Note provided a commitment fee of \$45,000 and legal fees of \$3,000, which were also prepaid. If the commitment and legal fees were characterized as interest, the effective annual rate on the December Note, if repaid in full on May 3, 2012, would be 97.4%.

Duff & Phelps | Page 6 of 14

CCM has agreed, on a without prejudice basis, to exclude \$103,500 of the principal indebtedness owed to it by the Company under the December Note in its proposed credit bid. As a result, if the Stalking Horse Offer closes on May 3, 2012 and the December Note is considered to be repaid in full on that date, the effective annual rate of interest (with the commitment fee of \$45,000 and legal fees of \$3,000 being characterized as interest) would be 59.9%. Accordingly, the aggregate amount of the proposed credit bid pursuant to the Notes is \$3,296,500 (being the aggregate principal face value of the Notes (\$3,400,000) less \$103,500). A schedule reflecting these calculations is provided as Appendix "D".

CCM has advised the Receiver that should the Stalking Horse Offer not be the Successful Bid, it reserves all of its rights in connection with any distribution of the sale proceeds derived from an alternate transaction.

- Approximately \$157,000 of the October Note proceeds was applied to discharge certain of the Company's unsecured obligations owing to CCM at the time the October Note was entered into. This amount is included in CCM's proposed credit bid. In this regard, the Receiver notes that:
 - (i) the October Note and the general security agreement granted by the Company to CCM are each dated October 21, 2011. October 21, 2011 is outside the three month review period provided for under Section 95 of the BIA for review of transactions by arm's length creditors;
 - the Receiver is not aware of any evidence that the security was granted or the payment was made with the intention to defraud, defeat or delay a creditor;
 - (iii) the amount paid by the Company to repay its unsecured indebtedness to CCM and to be included in the proposed credit bid represents approximately 5% of the aggregate amount of the proposed credit bid; and
 - (iv) the Receiver's counsel has conducted a litigation search in Toronto, Ontario, Mississauga, Ontario, the Province of Alberta (where the Company was originally incorporated) and the Federal Court of Canada and has included on the Service List any party identified as a plaintiff against the Company or a codefendant of the Company in any proceeding revealed by the litigation searches. These "litigation parties" will be provided a copy of this Report and notice of the motion to approve CCM's Stalking Horse Offer.

Duff & Phelps | Page 7 of 14

4.1 Security Opinion

Immediately following its appointment, the Receiver instructed Blakes to render the Opinion. The Opinion was required expeditiously in this context given CCM's intentions to advance a stalking horse bid in the form of a "credit bid".

Subject to the standard assumptions and qualifications contained in the Opinion, Blakes is of the view that the security granted by the Company in favour of CCM, as registered pursuant to the PPSA, creates a valid and perfected security interest in the business and assets of the Company. The Opinion does not speak to the quantum of CCM's secured debt - those issues will be addressed in the factum to be filed by Blakes on the Receiver's behalf.

5.0 The Stalking Horse Offer

On March 3, 2012, CCM submitted the Stalking Horse Offer to the Receiver. The Stalking Horse Offer is in the form of an Asset Purchase Agreement between CCM and the Receiver. In the days following its submission, the Receiver, CCM and their respective legal counsel negotiated the terms and provisions of the Stalking Horse Offer, a summary of which is as follows:

- CCM would acquire substantially all of the Company's business and Assets;
- The purchase price is equal to the Assumed Liabilities (as defined in the Stalking Horse Offer), plus CCM's secured debt outstanding: (i) under the Notes (reduced as set out in section 4 above); (ii) the costs of seeking and obtaining the Receivership Order (which are provided for in paragraph 30 of the Receivership Order); and (iii) under the Term Sheet as evidenced by Receiver's Certificates (collectively, the "Purchase Price"). The Purchase Price is estimated to be up to approximately \$3,744,000 (comprised of \$3,296,500 under the Notes, approximately \$47,500 for CCM's costs to bring the receivership application and obtain the Receivership Order and \$400,000 under the Term Sheet), before the value of Assumed Liabilities;
- It is a condition of closing that the Receiver shall have received sufficient funds
 to satisfy all obligations secured by the Receiver's Charge and the Receiver's
 Borrowings Charge (each as defined in the Receivership Order and both of
 which rank in priority to the Notes), either by way of Receiver's borrowings or
 through cash paid by the Purchaser, in either case to be added to the Purchase
 Price:
- CCM will assume certain liabilities, including any obligations incurred by the Receiver from operating the Company's business during the receivership proceedings, obligations under contracts assigned to CCM, obligations arising after the closing of the transaction and employee obligations with respect to transferred employees. CCM intends to offer employment to all of the Company's current employees on terms no less favourable to the employees should it be the "Successful Bidder" under the Sale Process, except that CCM will only recognize prior service of employees to the extent required by law;

Duff & Phelps | Page 8 of 14

- The Stalking Horse Offer is in the form of a "credit bid". Accordingly, it
 contemplates that the Purchase Price be satisfied by delivering releases and
 waivers under the Notes and the Term Sheet and with respect to the costs of
 seeking and obtaining the Receivership Order;
- The Stalking Horse Offer is consistent with standard insolvency transactions, i.e. completed on an "as is, where is" basis with only basic market standard representations or warranties;
- The closing date is to be as soon as practicable following satisfaction or waiver
 of all conditions to closing and the Purchaser is to use commercially
 reasonable efforts to satisfy all conditions to closing by May 3, 2012, if it is the
 Successful Bidder;
- The Stalking Horse Offer contemplates that May 15, 2012 would be the "Termination Date", being the date on which the Stalking Horse Offer could be terminated should the transaction not close prior to that date;
- The Stalking Horse Offer is subject to the approval of this Court, representing the only material condition precedent to the transaction; and
- The Stalking Horse Offer contemplates Court approval of the Sale Process and Bidding Procedures, as detailed in Section 6 below.

In the Receiver's view, the terms of the Stalking Horse Offer are generally consistent with Canadian insolvency transactions. A copy of the Stalking Horse Offer is attached as Appendix "E".

6.0 Sale Process and Bidding Procedures⁷

A summary of the proposed Sale Process is as follows:

- The Receiver shall distribute to prospective purchasers identified by the Receiver and the Company a brief interest solicitation letter detailing this opportunity. Attached to the interest solicitation letter will be a form of confidentiality agreement ("CA"). The CA may include additional protection for particularly sensitive information in the hands of the Company's competitors;
- The Receiver shall advertise the acquisition opportunity in *The Globe and Mail* (National Edition);

Duff & Phelps | Page 9 of 14

⁷ Capitalized terms not otherwise defined in this section have the meanings ascribed to them in the Bidding Procedures (Schedule D to the Stalking Horse Offer).

- The Receiver is in the process of preparing a confidential information memorandum ("CIM") that provides an overview of the Company's business, assets and financial results. The CIM will be made available to parties that execute a CA;
- Upon execution of a CA, prospective bidders will be provided with the
 opportunity to commence due diligence, including reviewing information in an
 online data room to be maintained by the Receiver. The data room will include,
 among other things, a summary of the Company's research and development
 activities, expenditures and results, the Company's patents and other relevant
 information on the Company's intellectual property, and its historical audited
 and unaudited financial statements;
- The Receiver will facilitate due diligence efforts by, *inter alia*, arranging site visits and meetings between key employees and interested parties, provided that such meetings are supervised by the Receiver, and the Receiver is of the view that such prospective purchasers are *bona fide*, in the Receiver's sole discretion; and
- Prospective purchasers will be provided with a copy of the Stalking Horse Offer. Prospective purchasers will be required to submit offers in the form of the Stalking Horse Offer.

In carrying out the Sale Process, none of the details of the Sale Process that would not otherwise be made available to other prospective purchasers, including the identity and/or number of parties participating in the process, will be disclosed by the Receiver to CCM or its legal counsel.

The Receiver is seeking approval of the proposed Bidding Procedures, which are attached as Schedule "D" to the Stalking Horse Offer. The Bidding Procedures are summarized as follows:

- Offers will be required to be submitted to the Receiver by 10:00 am (Toronto time) on April 16, 2012 (the "Bid Deadline"), being one month from the return date of the Receiver's motion for approval of the Sale Process and Bidding Procedures:
- For a Bidder to be considered a Qualified Bidder, the Bidder's offer will have to provide each of the following on or before the Bid Deadline:
 - (i) identification of the Bidder and any principals, and the representatives thereof;
 - (ii) written evidence of an appropriate senior executive's approval of the contemplated transaction and if the Bidder is an entity specially formed for the purpose of effectuating the contemplated transaction, then the Bidder must furnish approval of the contemplated transaction by the equity holder(s) of such Bidder and any guarantor(s) of the Bid;

Duff & Phelps | Page 10 of 14

- (iii) an executed CA; and
- (iv) written evidence to determine whether the Bidder has the financial wherewithal to complete the contemplated transaction.

A Bid must:

- (i) be in the form of an executed asset purchase agreement with marked revisions to the Stalking Horse Offer;
- (ii) be irrevocable until the earlier of (a) the day on which the Bidder is notified that the Bid is not a Qualified Bid; (b) the day on which the Successful Bid is selected, if the Bid is neither the Successful Bid nor the Back-Up Bid selected on such day; and (c) the Termination Date of May 15, 2012;
- (iii) not be subject to any diligence, internal approval, financing or other conditions that are more burdensome than the conditions in the Stalking Horse Offer;
- (iv) not contemplate a break fee, expense reimbursement provision or the like:
- (v) be accompanied by a cash deposit of not less than 15% of the Bid;
- (vi) contemplate a purchase price equal to or greater than the estimated credit bid amount of \$3,744,000, plus the Minimum Overbid Increment of \$100,000, plus the maximum amount of the Expense Reimbursement of \$75,000, for total consideration of at least \$3,919,000; and
- (vii) contain certain other requirements provided for in the Bidding Procedures.
- If no Qualified Bids are submitted by the Bid Deadline, the Stalking Horse Offer shall be accepted, subject to Court approval;
- If one or more Qualified Bids are received by the Bid Deadline, the Receiver shall conduct an Auction on April 20, 2012 at 10:00 AM (Toronto time) to determine the highest and/or best Bid with respect to the Assets;
- Bidding at the Auction shall be restricted to Qualified Bidders, which includes the Stalking Horse. Bidding at the Auction shall be conducted in rounds. The highest Qualified Bid at the beginning of the Auction shall constitute the Opening Bid and the highest Overbid at the end of each round shall constitute the Opening Bid for the following round. If at the end of any round of bidding, a Qualified Bidder (other than the Qualified Bidder that submitted the Opening Bid for such round), did not submit an Overbid, then the Receiver may, in its sole discretion, bar such Qualified Bidder from participating in the next round of bidding;

Duff & Phelps | Page 11 of 14

- The Receiver, with the assistance of its advisors, will determine the Opening Bid for each round:
- An Overbid must be for the purchase price of the Opening Bid of a round, plus the Minimum Overbid Increment of \$100,000. If the Opening Bid of a round is submitted by CCM, an Overbid must be for the purchase price of such bid, plus the Minimum Overbid Increment of \$100,000, plus the maximum Expense Reimbursement of \$75,000;
- If, in any round of bidding, no new Overbid is made, the Auction shall be closed and the Receiver shall declare the last Opening Bid as the Successful Bid and the second highest bid as the Back-Up Bid;
- The Successful Bidder and the Back-Up Bidder shall each, within two Business
 Days of the conclusion of the Auction, provide the Receiver with an additional
 cash deposit equal to fifteen percent (15%) of the total cash purchase price
 contemplated by the Successful Bid or the Back-Up Bid, as applicable;
- The Receiver shall, within two Business Days of the conclusion of the Auction, return all cash deposits received from Bidders other than the Successful Bidder and the Back-Up Bidder;
- The Receiver shall, within seven days of the conclusion of the Auction, or if there is no Auction, on April 20, 2012, serve notice of a Sale Hearing to approve the sale of the Purchased Assets to the Successful Bidder; and
- If the Successful Bid is consummated and the Successful Bidder is not CCM, the Expense Reimbursement, in the maximum amount of \$75,000, shall be paid by the Receiver to CCM from the proceeds of sale of the Successful Bid.

As noted above, the detailed Bidding Procedures are a separate schedule to the Stalking Horse Offer (Schedule "D"), which is appended to this Report as Appendix "E".

6.1 Sale Process Recommendation

The Receiver recommends that this Court issue an order approving the Sale Process and Bidding Procedures for the following reasons:

- In the Receiver's view, the Sale Process is commercially reasonable and the Bidding Procedures and Auction, if required, provide an opportunity for a result superior to the transaction contemplated by the Stalking Horse Offer;
- In the Receiver's view, the one month Sale Process is sufficient to allow interested parties to perform diligence and to submit offers. The Receiver does not have access to sufficient funding to support operations during a lengthy sale process. The Receiver believes a focused, transparent and expedited process is in the best interest of all stakeholders;

Duff & Phelps | Page 12 of 14

- The duration of the proposed Sale Process and the existence of a "stalking horse" offer should assist to create certainty for all stakeholders, particularly the Company's employees and potential customers. In this regard, the Company is in its development stage and certain customer trials are ongoing. Accordingly, it is important that the Sale Process be completed expeditiously so that stakeholders understand that the Company is to continue to operate as a going concern; and
- The use of a "stalking horse" offer, together with the Expense Reimbursement mechanism, has been used on many occasions in Canadian insolvency proceedings. The Expense Reimbursement is limited to a maximum of \$75,000, representing a maximum of approximately 2% of the value of the Stalking Horse Offer, which in the Receiver's view is reasonable and consistent with amounts used generally in Canadian insolvency proceedings. In the Receiver's view, the Expense Reimbursement will not discourage a third party from submitting an offer that is superior to the Stalking Horse Offer. In addition, there is no break fee or similar type payment contemplated by the Stalking Horse Offer.

7.0 Overview of the Receiver's Activities

The Receiver is requesting approval of its activities since the commencement of these proceedings, including the following:

- Reviewing and commenting on all Court materials filed in the context of the February 28th receivership application;
- Negotiating the Term Sheet with CCM;
- Carrying out the Receiver's duties and responsibilities in accordance with the Receivership Order, including overseeing the Company's operations;
- Opening receivership bank accounts and transferring funds from the Company's accounts in accordance with the Receivership Order;
- Taking possession of the Company's safety deposit box and its contents;
- Dealing with employee issues, including terminating, on the Company's behalf, certain employees in accordance with the Receivership Order;
- Attending at the Company's premises and convening employee meetings immediately following its appointment;
- Reviewing and commenting on a draft press release issued on February 29, 2012;

Duff & Phelps | Page 13 of 14

- Dealing with cash management issues, including paying post-filing expenses from the receivership accounts, including payroll;
- Issuing Receiver's Certificates for advances made by CCM under the Term Sheet in accordance with the Receivership Order;
- Arranging for the back-up of the Company's electronic records;
- Corresponding extensively with key stakeholders in these proceedings, including CCM, employees, suppliers, shareholders, certain parties asserting proprietary claims in intellectual property used by the Company and/or their respective legal counsel;
- Negotiating the Stalking Horse Offer and the Bidding Procedures;
- Drafting Sale Process materials;
- Corresponding with Osler Hoskin & Harcourt LLP, CCM's legal counsel, and Blakes, regarding various matters in respect of this mandate;
- Placing on its website copies of materials filed in these proceedings;
- Corresponding with Bennett Jones LLP, the Company's intellectual property legal counsel, regarding various matters in respect of the Company's intellectual property;
- Drafting this Report; and
- Addressing all other matters pertaining to the administration of these receivership proceedings.

8.0 Conclusion and Recommendation

Based on the foregoing, the Receiver respectfully recommends that this Court make an order granting the relief detailed in Section 1.1 of this Report.

All of which is respectfully submitted,

Duff & Phelm Canada Restructioning Inc.

DUFF & PHELPS CANADA RESTRUCTURING INC.
IN ITS CAPACITY AS COURT-APPOINTED RECEIVER OF

BLUTIP POWER TECHNOLOGIES LTD.

AND NOT IN ITS PERSONAL CAPACITY

Duff & Phelps | Page 14 of 14

Appendix "A"

Court File No.

C17-12-9622-0056

ONTARIO SUPERIOR COURT OF JUSTICE COMMERCIAL LIST

THE HONOURABLE MR.)	TUESDAY, THE 28 th DAY
JUSTICE MORAWETZ)	OF FEBRUARY, 2012

CCM MASTER QUALIFIED FUND, LTD.

Applicant

- and -

BLUTIP POWER TECHNOLOGIES LTD.

Respondent

APPOINTMENT ORDER

THIS APPLICATION made by the Applicant for an Order pursuant to section 243(1) of the *Bankruptcy and Insolvency Act*, R.S.C. 1985, c. B-3, as amended (the "BIA") appointing Duff & Phelps Canada Restructuring Inc. as receiver (in such capacity, the "Receiver") without security, of all of the assets, undertakings and properties of Blutip Power Technologies Ltd. (the "Debtor") acquired for, or used in relation to a business carried on by the Debtor, was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the affidavit of James Schuler sworn February 27, 2012 and the Exhibits thereto (the "Schuler Affidavit"), and on hearing the submissions of counsel for the Applicant and the proposed Receiver, no other parties appearing although duly served as appears from the affidavit of service of Arlene Mack sworn February 27, 2012 and the affidavit of service of Kevin MacEachern sworn February 27, 2012 and on reading the consent of Duff & Phelps to act as the Receiver,

SERVICE

1. **THIS COURT ORDERS** that the time for service of the Notice of Motion and the Motion is hereby abridged and validated so that this motion is properly returnable today and hereby dispenses with further service thereof.

APPOINTMENT

2. **THIS COURT ORDERS** that pursuant to section 243(1) of the BIA, Duff & Phelps is hereby appointed Receiver, without security, of all of the assets, undertakings and properties of the Debtor acquired for, or used in relation to a business carried on by the Debtor, including all proceeds thereof (the "**Property**").

RECEIVER'S POWERS

- 3. THIS COURT ORDERS that the Receiver is hereby empowered and authorized, but not obligated, to act at once in respect of the Property and, without in any way limiting the generality of the foregoing, the Receiver is hereby expressly empowered and authorized to do any of the following where the Receiver considers it necessary or desirable:
 - (a) to take possession of and exercise control over the Property and any and all proceeds, receipts and disbursements arising out of or from the Property;
 - (b) to receive, preserve, and protect the Property, or any part or parts thereof, including, but not limited to, the changing of locks and security codes, the relocating of Property to safeguard it, the engaging of independent security personnel, the taking of physical inventories and the placement of such insurance coverage as may be necessary or desirable;
 - (c) to manage, operate, and carry on the business of the Debtor, including the powers to enter into any agreements, incur any obligations in the ordinary

course of business, cease to carry on all or any part of the business, or cease to perform any contracts of the Debtor;

- (d) to engage consultants, appraisers, agents, experts, auditors, accountants, managers, counsel and such other persons from time to time and on whatever basis, including on a temporary basis, to assist with the exercise of the Receiver's powers and duties, including without limitation those conferred by this Order;
- (e) to purchase or lease such machinery, equipment, inventories, supplies, premises or other assets to continue the business of the Debtor or any part or parts thereof;
- (f) to receive and collect all monies and accounts now owed or hereafter owing to the Debtor and to exercise all remedies of the Debtor in collecting such monies, including, without limitation, to enforce any security held by the Debtor;
- (g) to settle, extend or compromise any indebtedness owing to the Debtor;
- (h) to execute, assign, issue and endorse documents of whatever nature in respect of any of the Property, whether in the Receiver's name or in the name and on behalf of the Debtor, for any purpose pursuant to this Order;
- (i) to initiate, prosecute and continue the prosecution of any and all proceedings and to defend all proceedings now pending or hereafter instituted with respect to the Debtor, the Property or the Receiver, and to settle or compromise any such proceedings. The authority hereby conveyed shall extend to such appeals or applications for judicial review in respect of any order or judgment pronounced in any such proceeding;
- (j) to market any or all of the Property, including advertising and soliciting offers in respect of the Property or any part or parts thereof and

negotiating such terms and conditions of sale as the Receiver in its discretion may deem appropriate;

- (k) to sell, convey, transfer, lease or assign the Property or any part or parts thereof out of the ordinary course of business,
 - (i) without the approval of this Court in respect of any transaction not exceeding \$25,000, provided that the aggregate consideration for all such transactions does not exceed \$150,000; and
 - (ii) with the approval of this Court in respect of any transaction in which the purchase price or the aggregate purchase price exceeds the applicable amount set out in the preceding clause;

and in each such case notice under subsection 63(4) of the Ontario *Personal Property Security Act*, or section 31 of the Ontario *Mortgages Act*, as the case may be, shall not be required, and in each case the Ontario *Bulk Sales Act* shall not apply;

- (1) to apply for any vesting order or other orders necessary to convey the Property or any part or parts thereof to a purchaser or purchasers thereof, free and clear of any liens or encumbrances affecting such Property;
- (m) to report to, meet with and discuss with such affected Persons (as defined below) as the Receiver deems appropriate on all matters relating to the Property and the receivership, and to share information, subject to such terms as to confidentiality as the Receiver deems advisable;
- (n) to register a copy of this Order and any other Orders in respect of the Property against title to any of the Property;
- (o) to apply for any permits, licences, approvals or permissions as may be required by any governmental authority and any renewals thereof for and on behalf of and, if thought desirable by the Receiver, in the name of the Debtor;

- (p) to enter into agreements with any trustee in bankruptcy appointed in respect of the Debtor, including, without limiting the generality of the foregoing, the ability to enter into occupation agreements for any property owned or leased by the Debtor;
- (q) to exercise any shareholder, partnership, joint venture or other rights which the Debtor may have; and
- (r) to take any steps reasonably incidental to the exercise of these powers or the performance of any statutory obligations.

and in each case where the Receiver takes any such actions or steps, it shall be exclusively authorized and empowered to do so, to the exclusion of all other Persons (as defined below), including the Debtor, and without interference from any other Person.

DUTY TO PROVIDE ACCESS AND CO-OPERATION TO THE RECEIVER

- 4. THIS COURT ORDERS that (i) the Debtor, (ii) all of its current and former directors, officers, employees, agents, accountants, legal counsel and shareholders, and all other persons acting on its instructions or behalf, and (iii) all other individuals, firms, corporations, governmental bodies or agencies, or other entities having notice of this Order (all of the foregoing, collectively, being "Persons" and each being a "Person") shall forthwith advise the Receiver of the existence of any Property in such Person's possession or control, shall grant immediate and continued access to the Property to the Receiver, and shall deliver all such Property to the Receiver upon the Receiver's request.
- the existence of any books, documents, securities, contracts, orders, corporate and accounting records, and any other papers, records and information of any kind related to the business or affairs of the Debtor, and any computer programs, computer tapes, computer disks, or other data storage media containing any such information (the foregoing, collectively, the "Records") in that Person's possession or control, and shall provide to the Receiver or permit the Receiver to make, retain and take away copies thereof and grant to the Receiver unfettered access to and use of accounting, computer, software and physical facilities relating thereto, provided however that

nothing in this paragraph 5 or in paragraph 6 of this Order shall require the delivery of Records, or the granting of access to Records, which may not be disclosed or provided to the Receiver due to the privilege attaching to solicitor-client communication or due to statutory provisions prohibiting such disclosure.

6. THIS COURT ORDERS that if any Records are stored or otherwise contained on a computer or other electronic system of information storage, whether by independent service provider or otherwise, all Persons in possession or control of such Records shall forthwith give unfettered access to the Receiver for the purpose of allowing the Receiver to recover and fully copy all of the information contained therein whether by way of printing the information onto paper or making copies of computer disks or such other manner of retrieving and copying the information as the Receiver in its discretion deems expedient, and shall not alter, erase or destroy any Records without the prior written consent of the Receiver. Further, for the purposes of this paragraph, all Persons shall provide the Receiver with all such assistance in gaining immediate access to the information in the Records as the Receiver may in its discretion require including providing the Receiver with instructions on the use of any computer or other system and providing the Receiver with any and all access codes, account names and account numbers that may be required to gain access to the information.

NO PROCEEDINGS AGAINST THE RECEIVER

7. **THIS COURT ORDERS** that no proceeding or enforcement process in any court or tribunal (each, a "**Proceeding**"), shall be commenced or continued against the Receiver except with the written consent of the Receiver or with leave of this Court.

NO PROCEEDINGS AGAINST THE DEBTOR OR THE PROPERTY

8. THIS COURT ORDERS that no Proceeding against or in respect of the Debtor or the Property shall be commenced or continued except with the written consent of the Receiver or with leave of this Court and any and all Proceedings currently under way against or in respect of the Debtor or the Property are hereby stayed and suspended pending further Order of this Court.

NO EXERCISE OF RIGHTS OR REMEDIES

9. THIS COURT ORDERS that all rights and remedies against the Debtor, the Receiver, or affecting the Property, are hereby stayed and suspended except with the written consent of the Receiver or leave of this Court, provided however that this stay and suspension does not apply in respect of any "eligible financial contract" as defined in the BIA, and further provided that nothing in this paragraph shall (i) empower the Receiver or the Debtor to carry on any business which the Debtor is not lawfully entitled to carry on, (ii) exempt the Receiver or the Debtor from compliance with statutory or regulatory provisions relating to health, safety or the environment, (iii) prevent the filing of any registration to preserve or perfect a security interest, or (iv) prevent the registration of a claim for lien.

NO INTERFERENCE WITH THE RECEIVER

10. THIS COURT ORDERS that no Person shall discontinue, fail to honour, alter, interfere with, repudiate, terminate or cease to perform any right, renewal right, contract, agreement, licence or permit in favour of or held by the Debtor, without written consent of the Receiver or leave of this Court.

CONTINUATION OF SERVICES

11. THIS COURT ORDERS that all Persons having oral or written agreements with the Debtor or statutory or regulatory mandates for the supply of goods and/or services, including without limitation, all computer software, communication and other data services, centralized banking services, payroll services, insurance, transportation services, utility or other services to the Debtor are hereby restrained until further Order of this Court from discontinuing, altering, interfering with or terminating the supply of such goods or services as may be required by the Receiver, and that the Receiver shall be entitled to the continued use of the Debtor's current telephone numbers, facsimile numbers, internet addresses and domain names, provided in each case that the normal prices or charges for all such goods or services received after the date of this Order are paid by the Receiver in accordance with normal payment practices of the Debtor or such other practices as may be agreed upon by the supplier or service provider and the Receiver, or as may be ordered by this Court.

RECEIVER TO HOLD FUNDS

12. THIS COURT ORDERS that all funds, monies, cheques, instruments, and other forms of payments received or collected by the Receiver from and after the making of this Order from any source whatsoever, including without limitation the sale of all or any of the Property and the collection of any accounts receivable in whole or in part, whether in existence on the date of this Order or hereafter coming into existence, shall be deposited into one or more new accounts to be opened by the Receiver (the "Post Receivership Accounts") and the monies standing to the credit of such Post Receivership Accounts from time to time, net of any disbursements provided for herein, shall be held by the Receiver to be paid in accordance with the terms of this Order or any further Order of this Court.

EMPLOYEES

THIS COURT ORDERS that all employees of the Debtor shall remain the employees of the Debtor until such time as the Receiver, on the Debtor's behalf, may terminate the employment of such employees. The Receiver shall not be liable for any employee-related liabilities, including any successor employer liabilities as provided for in section 14.06(1.2) of the BIA, other than such amounts as the Receiver may specifically agree in writing to pay, or in respect of its obligations under sections 81.4(5) or 81.6(3) of the BIA or under the Wage Earner Protection Program Act.

PIPEDA

Information Protection and Electronic Documents Act, the Receiver shall disclose personal information of identifiable individuals to prospective purchasers or bidders for the Property and to their advisors, but only to the extent desirable or required to negotiate and attempt to complete one or more sales of the Property (each, a "Sale"). Each prospective purchaser or bidder to whom such personal information is disclosed shall maintain and protect the privacy of such information and limit the use of such information to its evaluation of the Sale, and if it does not complete a Sale, shall return all such information to the Receiver, or in the alternative destroy all such information. The purchaser of any Property shall be entitled to continue to use the personal

information provided to it, and related to the Property purchased, in a manner which is in all material respects identical to the prior use of such information by the Debtor, and shall return all other personal information to the Receiver, or ensure that all other personal information is destroyed.

LIMITATION ON ENVIRONMENTAL LIABILITIES

THIS COURT ORDERS that nothing herein contained shall require the 15. Receiver to occupy or to take control, care, charge, possession or management (separately and/or collectively, "Possession") of any of the Property that might be environmentally contaminated, might be a pollutant or a contaminant, or might cause or contribute to a spill, discharge, release or deposit of a substance contrary to any federal, provincial or other law respecting the protection, conservation, enhancement, remediation or rehabilitation of the environment or relating to the disposal of waste or other contamination including, without limitation, the Canadian Environmental Protection Act, the Ontario Environmental Protection Act, the Ontario Water Resources Act, or the Ontario Occupational Health and Safety Act and regulations thereunder (the "Environmental Legislation"), provided however that nothing herein shall exempt the Receiver from any duty to report or make disclosure imposed by applicable Environmental Legislation. The Receiver shall not, as a result of this Order or anything done in pursuance of the Receiver's duties and powers under this Order, be deemed to be in Possession of any of the Property within the meaning of any Environmental Legislation, unless it is actually in possession.

LIMITATION ON THE RECEIVER'S LIABILITY

16. THIS COURT ORDERS that the Receiver shall incur no liability or obligation as a result of its appointment or the carrying out the provisions of this Order, save and except for any gross negligence or wilful misconduct on its part, or in respect of its obligations under sections 81.4(5) or 81.6(3) of the BIA or under the Wage Earner Protection Program Act. Nothing in this Order shall derogate from the protections afforded the Receiver by section 14.06 of the BIA or by any other applicable legislation.

RECEIVER'S ACCOUNTS

- THIS COURT ORDERS that the Receiver and counsel to the Receiver shall be paid their reasonable fees and disbursements, in each case at their standard rates and charges unless otherwise ordered by the Court on the passing of accounts, and that the Receiver and counsel to the Receiver shall be entitled to and are hereby granted a charge (the "Receiver's Charge") on the Property, as security for such fees and disbursements, both before and after the making of this Order in respect of these proceedings, and that the Receiver's Charge shall form a first charge on the Property in priority to all security interests, trusts, liens, charges and encumbrances, statutory or otherwise, in favour of any Person, except for (i) any validly perfected purchase money security interests and (ii) any statutory encumbrance existing as at the date of this Order in favour of any Person which is a "secured creditor" as defined in the BIA in respect of any amounts that are subject to a super-priority claim under the BIA, and subject to sections 14.06(7), 81.4(4), and 81.6(2) of the BIA.
- 18. THIS COURT ORDERS that, at the request of the Applicant, any other party in interest or the Court, the Receiver and its legal counsel shall pass their accounts, and for this purpose the accounts of the Receiver and its legal counsel are hereby referred to a judge of the Commercial List of the Ontario Superior Court of Justice.
- 19. THIS COURT ORDERS that the Receiver shall be at liberty from time to time to apply reasonable amounts, out of the monies in its hands, against its fees and disbursements, including legal fees and disbursements, incurred at the standard rates and charges of the Receiver or its counsel, and such amounts shall constitute advances against their remuneration and disbursements.

FUNDING OF THE RECEIVERSHIP

20. THIS COURT ORDERS that the Receiver be at liberty and it is hereby empowered to borrow by way of a revolving credit or otherwise, such monies from time to time as it may consider necessary or desirable, including without limitation pursuant to the Term Sheet attached as Exhibit "B" hereto the Schuler Affidavit, provided that the outstanding principal amount does not exceed \$400,000 (or such greater amount as this Court may by further Order

authorize) at any time, at such rate or rates of interest as it deems advisable for such period or periods of time as it may arrange, for the purpose of funding the exercise of the powers and duties conferred upon the Receiver by this Order, including interim expenditures. The whole of the Property shall be and is hereby charged by way of a fixed and specific charge (the "Receiver's Borrowings Charge") as security for the payment of the monies borrowed, together with interest and charges thereon, in priority to all security interests, trusts, liens, charges and encumbrances, statutory or otherwise, in favour of any Person, but subordinate in priority to validly perfected purchase money security interests, any statutory encumbrance existing as at the date of this Order in favour of any Person which is a "secured creditor" as defined in the BIA in respect of any amounts that are subject to a super-priority claim under the BIA, the Receiver's Charge and the charges as set out in sections 14.06(7), 81.4(4), and 81.6(2) of the BIA.

- 21. **THIS COURT ORDERS** that neither the Receiver's Borrowings Charge nor any other security granted by the Receiver in connection with its borrowings under this Order shall be enforced without leave of this Court.
- 22. THIS COURT ORDERS that the Receiver is at liberty and authorized to issue certificates substantially in the form annexed as Schedule "A" hereto (the "Receiver's Certificates") for any amount borrowed by it pursuant to this Order.
- 23. **THIS COURT ORDERS** that the monies from time to time borrowed by the Receiver pursuant to this Order or any further order of this Court and any and all Receiver's Certificates evidencing the same or any part thereof shall rank on a *pari passu* basis, unless otherwise agreed to by the holders of any prior issued Receiver's Certificates.

SERVICE AND NOTICE

24. THIS COURT ORDERS that the Receiver be at liberty to serve this Order, any other materials and orders in these proceedings, any notices or other correspondence, by forwarding true copies thereof by prepaid ordinary mail, courier, personal delivery or electronic transmission to the Debtor's creditors or other interested parties at their respective addresses as last shown on the records of the Debtor and that any such service or notice by courier, personal delivery or electronic transmission shall be deemed to be received on the next business day

following the date of forwarding thereof, or if sent by ordinary mail, on the third business day after mailing.

25. THIS COURT ORDERS that the Applicant, the Receiver, and any party who has filed a Notice of Appearance may serve any court materials in these proceedings by emailing a PDF or other electronic copy of such materials to counsels' email addresses as recorded on the Service List from time to time, and the Receiver may post a copy of any or all such materials on its website at:

http://www.duffandphelps.com/services/investment_banking/Pages/Restructuring Cases.aspx.

GENERAL

- 26. **THIS COURT ORDERS** that the Receiver may from time to time apply to this Court for advice and directions in the discharge of its powers and duties hereunder.
- 27. **THIS COURT ORDERS** that nothing in this Order shall prevent the Receiver from acting as a trustee in bankruptcy of the Debtor.
- 28. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in the United States to give effect to this Order and to assist the Receiver and its agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Receiver, as an officer of this Court, as may be necessary or desirable to give effect to this Order or to assist the Receiver and its agents in carrying out the terms of this Order.
- 29. THIS COURT ORDERS that the Receiver be at liberty and is hereby authorized and empowered to apply to any court, tribunal, regulatory or administrative body, wherever located, for the recognition of this Order and for assistance in carrying out the terms of this Order, and that the Receiver is authorized and empowered to act as a representative in respect of the within proceedings for the purpose of having these proceedings recognized in a jurisdiction outside Canada.

- 30. THIS COURT ORDERS that the Applicant shall have its costs of this motion, up to and including entry and service of this Order, provided for by the terms of the Applicant's security or, if not so provided by the Applicant's security, then on a substantial indemnity basis to be paid by the Receiver from the Debtor's estate with such priority and at such time as this Court may determine.
- 31. THIS COURT ORDERS that any interested party may apply to this Court to vary or amend this Order on not less than seven (7) days' notice to the Receiver and to any other party likely to be affected by the order sought or upon such other notice, if any, as this Court may order.
- THIS COURT ORDERS that notwithstanding the immediately preceding paragraph, any motion to vary or amend the Receiver's Charge or the Receiver's Borrowings Charge must be brought and be returnable no later than March 20, 2012 and on not less than four (4) days' notice to the Receiver, the Applicant and any other party or parties likely to be affected by the order sought. The Receiver and any party which advances funds to the Receiver pursuant to paragraph 20 of this Order shall be entitled to rely on the priority granted to the Receiver's Charge and the Receiver's Borrowings Charge (as applicable) up to and including the day on which this Order is varied or modified.

ENTERED AT / INSCRIT A TORONTO

ON / BOOK NO: LE / DANS LE REGISTRE NO.:

FEB 2 8 2012

M

SCHEDULE "A"

RECEIVER CERTIFICATE

CERTIFICATE NO
AMOUNT \$
1. THIS IS TO CERTIFY that Duff & Phelps Canada Restructuring Inc., th
receiver (the "Receiver") of the assets, undertakings and properties Blutip Power Technologie
Ltd. acquired for, or used in relation to a business carried on by the Debtor, including a
proceeds thereof (collectively, the "Property") appointed by Order of the Ontario Superior
Court of Justice (Commercial List) (the "Court") dated the day of, 20 (the
"Order") made in an action having Court file numberCL, has received as suc
Receiver from the holder of this certificate (the "Lender") the principal sum of \$
being part of the total principal sum of \$ which the Receiver is authorized to
borrow under and pursuant to the Order.
2. The principal sum evidenced by this certificate is payable on demand by the
Lender with interest thereon calculated and compounded [daily][monthly not in advance on the
day of each month] after the date hereof at a notional rate per annum equal to the rate of
per cent above the prime commercial lending rate of Bank of from time t
time.
3. Such principal sum with interest thereon is, by the terms of the Order, together
with the principal sums and interest thereon of all other certificates issued by the Received
pursuant to the Order or to any further order of the Court, a charge upon the whole of th
Property, in priority to the security interests of any other person, but subject to the priority of th
charges set out in the Order and in the Bankruptcy and Insolvency Act, and the right of the
Receiver to indemnify itself out of such Property in respect of its remuneration and expenses.
4. All sums payable in respect of principal and interest under this certificate ar
payable at the main office of the Lender at Chicago, Illinois.

- 5. Until all liability in respect of this certificate has been terminated, no certificates creating charges ranking or purporting to rank in priority to this certificate shall be issued by the Receiver to any person other than the holder of this certificate without the prior written consent of the holder of this certificate.
- 6. The charge securing this certificate shall operate so as to permit the Receiver to deal with the Property as authorized by the Order and as authorized by any further or other order of the Court.

7.	The Receiver does not undertake, and it is not under any personal liability, to pay
any	sum in respect of which it may issue certificates under the terms of the Order.

DATED	the	day o	f	, 20	

DUFF & PHELPS CANADA RESTRUCTURING INC., solely in its capacity as Receiver of the Property, and not in its personal capacity

Per:		
	Name:	
	Title:	

Schedule "B"

TERM SHEET

February 27, 2012

Duff & Phelps Canada Restructuring Inc., in its capacity as Receiver of Blutip Power Technologies Ltd. 200 King Street West Suite 1002 Toronto, ON M5H 3T4

Attention:

David Sieradzki

CCM Master Qualified Fund, Ltd. (the "Lender") intends to make an application before the Ontario Superior Court of Justice (Commercial List) (the "Court") to seek the appointment of Duff & Phelps Canada Restructuring Inc. ("Duff & Phelps") as receiver over the assets, property and undertaking (collectively, the "Assets") of Blutip Power Technologies Ltd. (the "Company") pursuant to the terms of an order of the Court, substantially in the form of that annexed hereto as Schedule "A". Once Duff & Phelps is appointed as receiver (Duff & Phelps in such capacity, the "Receiver") the Receiver will require funding to pay certain costs and expenses incurred by the Company and the Receiver. The Lender is pleased to make the following credit facility (the "Credit Facility") available to Duff & Phelps, once it is appointed as Receiver, on and subject to the following terms and conditions:

AMOUNT:

CDN\$400,000.

PURPOSE:

To fund the powers and duties conferred upon the Receiver by the Order (as hereinafter defined), including managing, operating, and carrying on the business of the Company, and paying the reasonable fees and expenses of the Receiver and its legal counsel.

<u>USE OF PROCEEDS</u>: Proceeds of the Credit Facility will be used by the Receiver to finance general working capital needs and for general operating purposes of the Company deemed commercially reasonable by the Receiver during the receivership, including without limitation, the professional fees and costs of the Receiver and its counsel. For greater certainty, the Advances (as hereinafter defined) shall not be used by the Receiver or the Company to make payments to creditors, including any government authorities or agencies in respect of amounts due or accruing due to such creditors prior to the date of the Receivership Order, unless permitted by the Receivership Order or a subsequent order of the Court.

INTEREST RATE:

10% per annum (the "Interest Rate"), calculated and compounded monthly not in advance on the last day of each month. During the continuation of an Event of Default the Interest Rate shall be increased by 2% per annum.

AVAILABILITY

Advances under the Credit Facility ("Advances") shall be made to the Receiver in Canadian dollars only and by way of receiver's certificates ("Receiver's Certificates"), substantially in the form of the certificate attached as Schedule "A" to the Order. The Receiver shall not permit the sum of the aggregate principal amount of the Canadian Dollar value of all outstanding Advances under the Credit Facility to, at any time, exceed CDN\$400,000.

MATURITY DATE: The Credit Facility shall be repayable on April 30, 2012 or the day on which an Event of Default occurs, unless otherwise extended by written agreement of the Receiver and the Lender.

NO LIABILITY:

Duff & Phelps Canada Restructuring Inc. shall not have any personal liability to repay any principal amount or any interest, fee or other amount owing hereunder and the Lender's recourse with respect thereto shall be limited to the Assets and any proceeds thereof.

CONDITIONS PRECEDENT:

The Lender shall not be obliged to make any Advance under the Credit Facility unless the following conditions precedent have been satisfied or waived: (i) the Order has been issued by the Court; (ii) the Receiver has executed and returned a copy of this Term Sheet; (iii) the Receiver has delivered a Receiver's Certificate to the Lender by no later than 9:00 a.m. on the day that an Advance is to be made; and (iv) no Event of Default shall have occurred (collectively, the "Conditions Precedent").

SECURITY:

The Receiver's obligations under this Credit Facility shall be secured by the Receiver's Borrowing Charge as provided for in the Order and the Receiver's Certificates.

EVENTS OF DEFAULT

The occurrence of any one or more of the following events shall constitute an "Event of Default" under this Term Sheet:

- any termination of the stay of proceedings under the Order; (a)
- (b) failure by the Receiver to pay any principal amount outstanding hereunder when the same shall become due and payable hereunder;
- (c) any termination of the appointment of the Receiver;
- a final order is entered whereby the Order is reversed, stayed, (d) modified or amended without the express written consent of the Lender;

- (e) failure by the Receiver to pay any interest on any Advances made under the Credit Facility, any fees payable hereunder or any other amounts owing under the Facility when the same shall become due and payable; or
- if a sales process for the business of the Company and the Assets is not commenced by an order of the Court approving such sales process (the "Sale Process Order") within 2 weeks of the date of the Order (unless otherwise extended by written agreement of the Receiver and the Lender), which Sale Process Order shall be in form and substance satisfactory to the Lender and the Receiver, acting reasonably and which Sale Process Order shall not have been varied, amended or altered, without the express written consent of the Lender, not to be unreasonably withheld.

REMEDIES:

Upon the occurrence of an Event of Default and at any time thereafter, while an Event of Default is continuing, the Lender may declare, by giving notice to the Receiver, that the Credit Facility is terminated and cancelled and declare the Advances made to be immediately due and payable (whereupon the same shall become so payable, together with accrued interest thereon and any other sums then owed by the Receiver hereunder).

RECEIVER'S FEES: In addition to any amounts advanced under the Credit Facility and if required, the Lender agrees to pay the reasonable fees and expenses of the Receiver and those of its legal counsel, incurred at the standard rates and charges of the Receiver and its counsel in connection with their performance and duties under the Order and further orders of the Court made in the receivership proceedings of the Company; provided that in the event that the Credit Facility is terminated and cancelled upon the occurrence of an Event of Default as provided for herein (such date, the "Termination Date"), the Lender agrees to pay solely those such amounts that have been incurred to and including the Termination Date, even if such amounts are payable after the Termination Date. In the event that the Lender and the Receiver are unable to achieve a consensus as to the form and substance of the Sale Process Order, the Receiver shall be at liberty to apply to the Court for advice and directions, and the Lender agrees to pay the reasonable fees and expenses of the Receiver and those of its legal counsel, incurred at their standard rates and charges in connection with such motion for advice and directions.

GOVERNING LAW: This Term Sheet and each of the documents contemplated herein shall be governed by the laws of the Province of Ontario and the federal laws of Canada applicable therein.

<u>COUNTERPARTS</u>: This Term Sheet may be executed simultaneously in two or more counterparts, each of which shall be deemed an original, but all of which

together shall constitute one and the same instrument. Signatures provided by facsimile machine shall be valid and binding.

NOTICES:

Any notice or other communication to be delivered hereunder must be in writing and may, subject as hereinafter provided, be made or given by personal delivery, ordinary mail or by facsimile or email addressed to the respective parties as follows:

If to the Lender:

c/o Coghill Capital Management, LLC One North Wacker Drive, Suite 4350 Chicago, IL 60606

Attention:

James Schuler

Fax:

(312)324-2001

Email:

jschuler@coghillcapital.com

with a copy to:

Osler, Hoskin & Harcourt LLP 100 King Street West 1 First Canadian Place, Suite 6100 Toronto, Ontario M5X 1B8

Attention:

Marc Wasserman

Fax:

(416) 862-6666

Email:

mwasserman@osler.com

If to the Receiver

Duff & Phelps Canada Restructuring Inc. 200 King Street West Suite 1002 Toronto, Ontario M5H 3T4

Attention:

David Sieradzki

Fax:

(647) 497-9470

Email:

david.sieradzki@duffandphelps.com

with a copy to:

Blake Cassels & Graydon LLP 199 Bay Street Suite 4000, Commerce Court West Toronto, ON M5L 1A9 Attention:

Linc Rogers (416) 863-2653

Fax: Email:

linc.rogers@blakes.com

or to such other address as any party may from time to time notify the others in accordance with this Term Sheet. Any such communication so given or made shall be deemed to have been given or made and to have been received on the day of delivery if delivered, or on the day of faxing or sending by other means of recorded electronic communication, provided that such day in either event is a business day and the communication is so delivered, faxed or sent before 5:00 p.m. (Toronto time) on such day. Otherwise, such communication shall be deemed to have been given and made and to have been received on the next following Business Day.

[Remainder of Page Intentionally Left Blank]

The Lender's commitment to provide the Credit Facility on the terms set out herein will remain binding on the Lender and open for acceptance by the Receiver until 5:00 p.m. (Toronto time) on February 29, 2012. If the above terms and conditions are acceptable to you, please execute and return a copy of this Term Sheet by email no later than 5:00 p.m. (Toronto time) on February 29, 2012.

	COGHILL CAPITAL MANAGEMENT, LLC in its capacity as Investment Manager of CCM MASTER QUALIFIED FUND, LTD.		
	By: Name: Title:		
We acknowledge and accept the within terms and conditions this day of February, 2012.			
	DUFF & PHELPS CANADA RESTRUCTURING INC., in its capacity as Receiver of Blutip Power Technologies Ltd., without personal or corporate liability		
	By: Name: Title:		

SCHEDULE "A" FORM OF ORDER

(A)

•

Applicant

Respondent

ONTARIO SUPERIOR COURT OF JUSTICE (Commercial List)

Proceeding commenced at Toronto

APPOINTMENT ORDER

OSLER, HOSKIN & HARCOURT LLP P.O. Box 50, 1 First Canadian Place 100 King Street West Toronto, Canada M5X 1B8

Alexander F. Cobb (LSUC#45363F) (416) 862-5964

Marc S. Wasserman (LSUC#44066M) (416) 862-4908

Andrea M. Lockhart (LSUC#55444K) (416) 862-6829 (416) 862-6666

Lawyers for the Applicant

BLUTIP POWER TECHNOLOGIES LTD. and CCM MASTER QUALIFIED FUND, LTD.

Applicant

L. Rogers

Respondent

Court File No:

SUPERIOR COURT OF JUSTICE (Commercial List) ONTARIO

Proceeding commenced at Toronto

OSLER, HOSKIN & HARCOURT LLP P.O. Box 50, 1 First Canadian Place

Toronto, Canada M5X 1B8 100 King Street West

Alexander F. Cobb (LSUC#45363F) (416) 862-5964 Marc S. Wasserman (LSUC#44066M) (416) 862-4908 (416) 862-6666

Lawyers for the Applicant

7 243

APPLICATION RECORD

it county he so sound. Itelity
in the irenthous in required cal
this on he around that he appoint
of a Receive.
The Application is greated. The
and stall issue to the form
proseted as smalled. (It I sim Sheet
has here spreading and all
the Court is regard that leave of
the Court is regard to severe
renedict to the enex of default)

Alaris).

Appendix "B"

Court File No.

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

IN THE MATTER OF AN APPLICATION PURSUANT TO SECTION 243(1) OF THE *BANKRUPTCY AND INSOLVENCY ACT*, R.S.C. 1985, c. B-3, AS AMENDED

BETWEEN:

CCM MASTER QUALIFIED FUND, LTD.

Applicant

- and -

BLUTIP POWER TECHNOLOGIES LTD.

Respondent

AFFIDAVIT OF JAMES SCHULER (sworn February 27, 2012)

I, James Schuler, of the city of Naperville, in the State of Illinois, MAKE OATH AND SAY:

- 1. I am the Chief Financial Officer and a Principal of Coghill Capital Management, LLC ("Coghill"). Coghill is the investment manager of CCM Master Qualified Fund, Ltd. ("CCM"). I have been directly involved in managing the relationship between CCM and Blutip Power Technologies Ltd. ("Blutip"). As such, I have personal knowledge of the matters deposed to herein. Where I have relied on other sources for information, I have specifically referred to such sources and I believe them to be true.
- 2. I am swearing this affidavit in support of an application by CCM to appoint Duff & Phelps Canada Restructuring Inc. ("Duff & Phelps") as receiver (the "Receiver") over the

Collateral (as defined below) of Blutip pursuant to s. 243(1) of the *Bankruptcy and Insolvency Act* R.S.C., 1985, c. B-3, as amended (the "BIA").

Overview

- 3. Coghill is a Chicago-based investment advisor specializing in small cap value equity management. Pursuant to an agreement between Coghill and the CCM, Coghill acts as the manager of CCM's investments.
- 4. CCM is a corporation incorporated pursuant to the laws of the Cayman Islands. CCM's primary objective is to seek capital appreciation over the long-term by holding both long and short positions in small cap companies.
- Blutip is a publicly traded technology company based in Mississauga, Ontario that is focused on improving engine fuel economy and reducing fuel particulates through the use of certain proprietary, patented and patent-pending fuel optimization and emissions control systems. The company is still in its development phase and, in the absence of positive cash flow from operating activities, relies on external sources of funding.
- 6. To such end, CCM has supported Blutip over the course of more than 6 years through significant capital injections totalling approximately \$6.15 million. In particular, since January 2006 CCM has participated in all of Blutip's equity offerings either as lead investor or on a pro-rata basis.
- In addition, over the last few months CCM has provided emergency liquidity to Blutip on a secured basis. As described in greater detail below, Blutip is indebted to CCM in the amount of approximately \$3.4 million (together with fees, costs and other allowable charges which continue to accrue, the "Indebtedness") pursuant to two convertible senior secured promissory notes dated October 21, 2011 and December 29, 2011 (collectively, the "Notes"). The Indebtedness is secured by a General Security Agreement issued by Blutip in favour of CCM dated as of October 21, 2011 pursuant to which CCM holds comprehensive security over all of the property, assets and undertaking of Blutip (the "Collateral").
- 8. CCM is the senior secured creditor of Blutip, and to the best of my knowledge, by far its largest overall creditor. Events of Default have occurred under the Notes and such Events

of Default are continuing and remain uncured. The value of the Collateral securing the Indebtedness is at risk and Blutip is facing an immediate liquidity crisis and requires additional financing to meet its working capital needs. Among other things, (a) Blutip is unable to meet all of its obligations as they generally became due, including payroll obligations, (b) all of the directors and officers of Blutip (other than the VP & Corporate Secretary Shannon Brander (the "Corporate Secretary")) have resigned, and (c) trading in Blutip's securities has been suspended for failing to maintain exchange listing requirements, and the TSX Venture Exchange ("TSXV") notified Blutip by letter dated February 23, 2012 that the company would be reclassified as a Tier-2 company should the company fail to rectify such deficiency and/or the company's securities remain suspended after 10 business days. Such notice period expires March 12, 2012.

- 9. On February 27, 2012, Osler, Hoskin & Harcourt LLP ("Osler"), counsel for CCM, sent Blutip a letter demanding immediate repayment of the Indebtedness. On such date, Osler, on behalf of CCM, also sent Blutip a notice of intention to enforce security pursuant to s. 244 of the BIA and enclosed a consent to early enforcement to be signed at Blutip's option. CCM will advise the Court whether Blutip has executed the Consent at the hearing of this Application. In either case, I believe that a Receiver should be appointed without delay and, therefore, that the 10 day notice period contemplated by the BIA should be abridged.
- OCM is bringing this application to have Duff & Phelps appointed as Receiver over the Collateral of Blutip in order to, among other things, stabilize the business. Blutip has no directors or officers (other than the Corporate Secretary) and no money with which to fund urgent payroll and other operational requirements including occupancy costs and research and development activities. A Receiver is urgently required to preserve the value of the business as a going-concern, thereby protecting CCM's interest in the Collateral and its ability to recover amounts owing in respect of the Indebtedness. Subject to the Receiver's views and the Court's discretion, it is anticipated that the Receiver will bring a motion before this Honourable Court in the next few weeks to approve a marketing and sales process for the business. Blutip requires the stability and certainty of a Court-supervised receivership during this time in light of the continuing defaults and Blutip's critical financial position and liquidity crisis. Further, CCM is not prepared to provide additional financing to Blutip other than in connection with a court-supervised receivership process.

Blutip's Business

- Blutip is a control technology company that is focused on improving fuel economy and reducing opacity (particulates) through the use of certain proprietary, patented and patent-pending fuel optimization and emissions control systems.
- Blutip was incorporated on March 29, 2004 under the *Business Corporations Act* (Alberta) and continued under the *Business Corporations Act* (Ontario) on May 30, 2011, at which time it changed its name from Hy-Drive Technologies Ltd. to Blutip Power Technologies Ltd.
- 13. The company is headquartered and operates from leased premises located at 6705 Millcreek Drive, Suite 4, Mississauga, Ontario, and until recently as described herein, was publicly traded on the TSXV under the ticker-symbol "BPR". At present, CCM owns approximately 19.2% of the issued and outstanding shares of the company. In addition, CCM holds conversion privileges under the Notes and exercise rights under 36,500,000 warrants. However, the Notes and the warrants provide that CCM may not exercise such conversion privileges or exercise rights to the extent it would result in CCM (alone or in combination with any other person), directly or indirectly, holding, beneficially owning or controlling or directing more than 19.9% of the issued and outstanding common shares of Blutip. If CCM did exercise all of its conversion privileges and exercise rights, CCM's ownership percentage in Blutip would increase to approximately 57%. Of course, there is no current prospect of that happening because of the blocker provision and because conversion or the exercise of warrants at \$0.10 and \$0.46 per share would not be economical.
- 14. Blutip has 16 employees, who are not unionized. Blutip does not maintain any pension plans.
- At present, Blutip is still in a development phase with no significant sources of revenue. Pursuant to Blutip's most recently published consolidated unaudited financial statements for the three and nine months ended September 30, 2011, the company had a negative cash flow from operating activities of \$3,510,770 and a net and comprehensive loss of \$3,131,890 for the nine-months ended September 30, 2011. As at September 30, 2011, Blutip's balance sheet reflects a shareholders' deficit of \$64,906,158. As such, Blutip depends on external sources of funding to continue to operate as a going concern.

Indebtedness and Security

The Notes

- Blutip issued a Convertible Senior Secured Promissory Note to CCM dated October 21, 2011 (the "October Note") in the principal amount of \$2,600,000 ("October Note Principal Amount"). The proceeds of the October Note were to be used by Blutip to (i) repay the principal amount of accrued and unpaid interest on certain existing convertible secured notes in favour of third parties not related to CCM, (ii) repay the unsecured promissory notes issued by Blutip to CCM dated September 30, 2011 and October 14, 2011, (iii) pay interest on the October Note, (iv) pay the reasonable out-of-pocket expenses incurred by CCM in connection with the preparation, negotiation, execution, delivery and administration of the October Note and related documentation, (v) repay amounts advanced by CCM to Blutip prior to the date thereof, and (vi) for general working capital purposes in accordance with the sources and uses table appended thereto. Pursuant to section 3 of the October Note, the amounts set out in (i) to (v) above were deducted by CCM from the total loan proceeds advanced to Blutip on closing and paid by CCM, for and on behalf of Blutip, to the applicable payees. A copy of the October Note is appended as Exhibit "A" to this Affidavit.
- a Convertible Senior Secured Promissory Note issued by Blutip to CCM on December 29, 2011 (the "December Note") in the principal amount of \$800,000 (the "December Note Principal Amount). The proceeds of the December Note were to be used to (i) pay interest on the December Note due on the closing date, (ii) pay the reasonable out-of-pocket expenses incurred by CCM in connection with the preparation, negotiation, execution, delivery and administration of the December Note and related documentation, (iii) pay to CCM a commitment fee of \$45,000, and (iv) for general working capital purposes. Pursuant to section 3 of the December Note, the amounts set out in (i) to (iii) above were deducted by CCM from the total loan proceeds advanced to Blutip on closing and retained by CCM in satisfaction of such debts. A copy of the December Note is appended as Exhibit "B" to this Affidavit.
- 18. According to Blutip's press releases dated October 31, 2011 and December 29, 2011, due to CCM's shareholdings in Blutip, each of the Note offerings constituted a "related party transaction" pursuant to Multilateral Instrument 61-101 ("MI 61-101") published by the

Ontario Securities Commission. Each of the Notes was issued in reliance on a financial hardship exemption from the minority approval requirement of MI 61-101 for related transactions as, among other things, Blutip was in serious financial difficulty. Copies of the October 31 and December 29 press releases are appended as Exhibit "C" to this Affidavit.

Payment Terms

(a) October Note

- Pursuant to the October Note, Blutip promised to pay to the order of CCM, or its permitted assigns, the October Note Principal Amount, together with any accrued interest thereon on the earlier to occur of the following dates (the "October Note Maturity Date"): (i) the two year anniversary of the Effective Date (being the first business day following the date of receipt of the required regulatory approvals in connection with the issuance of the October Note), or (ii) when declared due and payable by CCM upon the occurrence of an Event of Default (as defined in the October Note) or otherwise deemed to be due and payable upon the occurrence of certain Events of Default.
- 20. The October Note bears interest at 15% per annum, calculated from the Effective Date to the latest October Note Maturity Date. During the occurrence and continuation of an Event of Default, the October Note bears additional default interest at the rate of 2% per annum, calculated daily and payable monthly in arrears on the last business day of each month. In accordance with the terms of the October Note, non-default interest was paid in advance on closing by deducting such interest (totalling \$780,000) from the total loan proceeds advanced by CCM to Blutip on closing as described above.
- 21. Pursuant to section 11 of the October Note, Blutip is prohibited from prepaying all or part of the October Note Principal prior to the October Note Maturity Date.

(b) December Note

22. The terms and conditions of the December Note are substantially similar to the terms and conditions of the October Note. Pursuant to the December Note, Blutip promised to pay to the order of CCM, or its permitted assigns, the December Note Principal Amount, together with any accrued interest thereon on the earlier to occur of the following dates (the "December Note Maturity Date"): (i) October 31, 2013 and (ii) when declared due and payable

by CCM upon the occurrence of an Event of Default (as defined in the December Note) or otherwise deemed to be due and payable upon the occurrence of certain Events of Default.

- 23. The December Note bears interest at 15% per annum, calculated from December 29, 2011 to the latest December Note Maturity Date. During the occurrence and continuation of an Event of Default, the December Note bears additional default interest at the rate of 2% per annum, calculated daily and payable monthly in arrears on the last business day of each month. In accordance with the terms of the December Note, non-default interest was paid in advance on closing by deducting such interest (totalling \$220,932) from the total loan proceeds advanced by CCM to Blutip on closing as described above.
- 24. Pursuant to section 11 of the December Note, Blutip is prohibited from prepaying all or part of the December Note Principal prior to the December Note Maturity Date.

Events of Default

- 25. Section 9 of each the Notes sets out a number Events of Default, including:
 - (a) if a material adverse change occurs in the business, property, assets, prospects, operations or condition, financial or otherwise, of Blutip;
 - (b) if Blutip becomes insolvent or generally unable to pay its debts as they become due or admits in writing its inability to pay its debts generally; and
 - (c) if the shares of Blutip cease to be listed on the TSXV, or the shares of Blutip are suspended from trading for a period in excess of 10 trading days.
- 26. Pursuant to section 10 of each of the Notes, upon the occurrence and during the continuance of an Event of Default: (i) CCM may require Blutip to immediately pay to CCM all amounts owing to CCM under the Notes (provided that all such amounts automatically become due and payable upon an insolvency Event of Default) and (ii) CCM may exercise any and all rights and remedies available to CCM pursuant to the Security or at law or in equity.

Security

27. Pursuant to a General Security Agreement granted by Blutip in favour of CCM dated as of October 21, 2011 (the "Security"), Blutip granted to CCM a first ranking security

interest in the Collateral, including accounts, chattel paper, documents, documents of title, contractual rights and insurance claims, intellectual property, leases and proceeds of the foregoing. Section 12(b) of the Security provides upon an Event of Default, which is continuing, CCM may make an application to the Court for the appointment of a receiver. A copy of the General Security Agreement is appended as Exhibit "D" to this Affidavit.

- 28. The Collateral secures all present and future obligations, indebtedness and liability of Blutip to CCM under the Notes and the Security. The security interest under the Security is registered under the *Personal Property Security Act* (Ontario) ("PPSA") under PPSA file reference number 673851519. A copy of the PPSA Searches dated February 12, 2012 against Blutip and its predecessor Hy-Drive Technologies Ltd. (the "PPSA Searches") is appended as Exhibit "E" to this Affidavit.
- The PPSA Searches reveal that there are no other PPSA security registrations against Blutip other than an all-asset security registration in favour of CCM and a registration over "accounts" and "other" in favour of Royal Bank of Canada ("RBC"). I understand from the Corporate Secretary that RBC's registration relates to a letter of credit that has subsequently expired. I am also advised by the Corporate Secretary that the proceeds of such letter of credit (\$25,000) were used to partially fund Blutip's payroll obligations due February 17, 2012, which totalled approximately \$55,000.
- 30. As set out below, Blutip has a number of trade creditors. To the best of my knowledge, none of those trade creditors are secured. As of February 21, 2012, the trade accounts payable totalled approximately \$287,000.

Defaults Occur under the Notes

- 31. As set out above, the Notes contain a number of Events of Default. Pursuant to the Notes, upon the occurrence of an Event of Default, CCM may, in addition to any other remedies under the Security or at law or equity, declare the Indebtedness to be immediately due and payable.
- 32. A number of events transpired over the course of February 2012 that have triggered Events of Default under the Notes and have raised CCM's concerns that the value of the Collateral securing the Indebtedness may be at risk.

- 33. The cash position report (the "February 2 Cash Position") delivered by the Corporate Secretary to CCM on February 2, 2012 demonstrated that Blutip's cash position has significantly deteriorated and that Blutip required approximately \$400,000 in additional funding for the month of February to meet its obligations as they become due, including critical supplier payments, payroll and benefits obligations, employer health tax and workers' compensation premiums and annual TSXV filing fees.
- In connection therewith, board member Peter Williams notified CCM on February 5, 2012 that Blutip was in a financially fragile state and that the board members had resigned or would be resigning. On February 2, 2012, Blutip announced that Dean Dussias had resigned as Director, Interim CEO, CFO and COO of Blutip effective immediately. On February 6, 2012, the Investment Industry Regulatory Organization of Canada announced trading in Blutip's securities had been suspended and Blutip announced that Peter Williams and Richard Marceau, Blutip's sole remaining directors, had tendered their resignations effective that day. Copies of the February 2 and February 6 press releases are appended as Exhibits "F" and "G", respectively, to this Affidavit.
- 35. On February 9, 2012, Blutip also announced that the TSXV had put it on notice for failing to maintain exchange listing requirements, trading would remain halted and its listing would be suspended on February 21, 2012. A copy of the February 9 press release is appended as Exhibit "H" to this Affidavit.

Subsequent Negotiations and The February 17 Order

- 36. As a result of the foregoing Events of Default, CCM has been in a position to make demand under the Notes and accelerate the Indebtedness for some time.
- On or about February 10, 2012, a group of investors approached CCM seeking CCM's participation in connection with a potential \$6 million private placement equity offering (the "Proposed Offering"). CCM was advised that the investors had retained Gowling Lafleur Henderson LLP as counsel in connection with the Proposed Offering. CCM was also advised that the investors would be bringing an application before the Court on February 17, 2012 to appoint an interim board of directors on an urgent basis. CCM was advised that the relief sought was necessary because Blutip was without a board of directors to approve the Proposed Offering

and that Blutip's exchange listing was going to be suspended by the TSXV on February 21, 2012.

- 38. Over the course of the next few days, CCM went to great lengths to co-operate with the potential investors in order to give Blutip a chance to achieve an out-of-court solution to its dire financial predicament. I had extensive discussions with the potential investors as to the basis on which they believed Blutip could succeed as a public company (including through the appointment of a new CEO and a qualified board and the provision of sufficient working capital funding). While these discussions were ongoing, CCM did not enforce its rights and remedies in respect of the Events of Default. CCM also consented to the appointment of the interim board.
- Pursuant to an Order of the Honourable Justice Newbould dated February 17, 2012 (the "February 17 Order"), the Court appointed Peter William, David Champion and Bob Lipic as the interim board of the company. A copy of the February 17 Order is appended as Exhibit "I" to this Affidavit.

The Interim Board Resigns and Funding Concerns Continue

- Blutip was advised by the Corporate Secretary on February 21 that Bob Lipic declined his board seat. On February 22, 2012, the remaining two interim directors of Blutip resigned. The transaction contemplated by the Proposed Offering was not completed.
- 41. As a result, Blutip has no directors or officers other than the Corporate Secretary and its liquidity crisis continues.
- The updated cash report provided by the Corporate Secretary to CCM on February 21, 2012 (the "February 21 Cash Position") demonstrates that Blutip's cash position has continued to erode and that it remains unable to meet its liabilities as they become due. As set out below, the February 21 Cash Position reflects that Blutip required approximately \$460,000 as at February 21, 2012 to fund its accounts payable and accrued liabilities.

	Feb 21/2012
Bank-CDN	\$28,470.14
Bank-USD	\$705.69
Cashable GIC	\$ -
Total Cash	\$29,175.83
A/P CDN	\$243,220.12
A/P USD	\$44,241.28
Purchase Clearing	\$1,711.92
Accrued Liability	\$210,658.18
Rent	\$ -
Salaries & Wages	\$ -
EHT Payable(Payroll)	\$1,011.56
Tax Payable (Payroll)	\$20,602.31
HST Receivable	\$(28,407.00)
Accounts Receivable	\$ -
Total A/P & Receivable	\$493,038.37
Cash Balance (Deficit)	\$(463,862.54)

Delivery of Demand Notice, Notice of Intention to Enforce a Security and Consent to Early Enforcement

- 43. CCM has had supported Blutip over the course of more than 6 years through significant capital injections and debt financing totalling approximately \$9.55 million. In particular, since January 2006 CCM has participated in all of Blutip's equity offerings either as lead investor or on a pro-rata basis. Despite CCM's substantial investment in and ongoing support of Blutip, Blutip has continued to experience financial difficulties. In light of the failure of the Proposed Offering and the resignation of the interim board, CCM has determined that it would be in the best interests of all of Blutip's stakeholders to immediately commence receivership proceedings and preserve the value of Blutip's business and assets as a going concern.
- Accordingly, by letter dated February 27, 2012 (the "**Demand Notice**"), Osler, on behalf of CCM, delivered a notice of default under the Notes to Blutip. Pursuant to the Demand Notice, Osler notified Blutip that one or more Events of Default had occurred including, without limitation:
 - (a) Blutip had become generally unable to pay its debts as they become due, as evidenced by, among other things the February 21 Cash Position, constituting an Event of Default under Section 9(h)(i) of the Notes;

- (b) the resignation of all of the members of the board and subsequent resignation of all of the members of the interim board constituted a material adverse change in the business, property, prospects, operation or condition of Blutip, and accordingly, an Event of Default under Section 9(f) of the Notes; and
- (c) the TSXV notice of failure to maintain listing requirements, suspension of trading and proposed Tier 2 reclassification also constituted a material adverse change in the business, property, prospects, operation or condition of Blutip, and accordingly, an Event of Default under Section 9(f) of the Notes.
- The Demand Notice further advised that, on behalf of CCM, Osler demanded payment of the existing Indebtedness, together with any additional fees, costs and other allowable charges accrued or accruing to the date thereof. A copy of the Demand Notice is appended as Exhibit "J" to this Affidavit.
- delivered a notice of intention to enforce security pursuant to s. 244 of the BIA (the "Section 244 Notice") and a consent to early enforcement of the Security to be executed at the option of Blutip (the "Consent"). CCM will advise the Court whether Blutip has executed the Consent at the hearing of this Application. In either case, I believe that a Receiver should be appointed without delay and, therefore, that the 10 day notice period contemplated by the BIA should be abridged. As set out herein, Blutip is out of money and has no functioning leadership. In my view, given Blutip's current condition and its lack of executive leadership, allowing Blutip to carry on as it is for a further 10 days will serve no purpose, and will in fact be extremely harmful. I also believe that there is a material risk that Blutip will not be able to fund its payroll obligations due February 29, 2012 absent the appointment of a Receiver. I therefore believe that it is imperative that a Receiver be appointed immediately. A copy of the Section 244 Notice is appended as Exhibits "K" to this Affidavit.

Proposed Appointment of a Receiver

The February 21 Cash Position demonstrates that Blutip does not have the requisite financial resources to meet its obligations as they become due, including the March 2, 2012 payroll of approximately \$55,000, without immediate funding. Under Blutip's agreement

with its payroll service provider, the payroll has to be funded by February 29, 2012. Absent receipt of additional funding or accounts receivable collections (the timing and quantum of which are unknown), Blutip has no other sources of funding. CCM is only prepared to fund Blutip's operations (including the February 29th payroll) on a priority basis in the context of a receivership proceeding. In this regard, the proposed receivership order contemplates that any advances made by CCM to the Receiver would be afforded a priority under the Receiver's Borrowing Charge (as defined in the draft Receivership Order). CCM has entered into a Term Sheet with the proposed Receiver to govern the terms and conditions of any such advances, a copy of which is appended as Exhibit "L" to this Affidavit.

- 48. The Receiver's Borrowing Charge in the amount of \$400,000 is CCM's estimate of operating costs (including payroll) for a 30 to 45 day period based on the historical financial information provided by Blutip to CCM from time to time, plus an estimate for the fees and disbursements of the Receiver and its counsel for such period.
- All of the directors and officers (other than the Corporate Secretary) of Blutip have resigned. Swift action is required to stabilize the business and provide Blutip's employees with comfort as to Blutip's ability to meet its employee obligations so as to retain necessary staff and preserve Blutip's operations as a going concern.
- 50. It is proposed that Duff & Phelps be appointed as Receiver over the Collateral. Duff & Phelps is a reputable corporate restructuring, financial advisory and consulting firm with extensive experience in restructuring transactions in Canada. I believe that the appointment of Duff & Phelps as Receiver will further the objectives of ensuring the fair and equitable treatment for Blutip's stakeholders. Duff & Phelps has consented to act as Receiver if appointed by this Honourable Court.

Proposed Sale Process

Subject to the granting of the requested relief by this Honourable Court, it is anticipated that Duff & Phelps would return to this Honourable Court within the next few weeks for approval of a marketing and sales process for Blutip's business and assets. It is anticipated that such marketing and sale process would include:

- (a) identification of prospective purchasers and preparation of a confidential information memorandum;
- (b) advertisement of the sale of Blutip's business and assets in a national newspaper;
- (c) facilitation of due diligence, including maintaining an online data room, facility tours and key employee interviews by prospective purchasers; and
- (d) preparation and negotiation of one or more agreements of purchase and sale.

52. In the circumstances, I believe the immediate appointment of a Receiver is necessary to properly and effectively take possession of, preserve, protect and allow for the orderly sale of Blutip's business and assets. As noted above, CCM is prepared to fund Blutip's operations on a priority basis and, accordingly, the proposed receivership proceeding also provides a mechanism in which Blutip's normal course operating costs can be funded, absent which it does not appear that such funding is available to Blutip.

SWORN BEFORE ME at the City of Chicago, in the State of Illinois on February 27, 2012.

Notary Public

James Schuler

OFFICIAL SEAL
JAMES DUGGAN
NOTARY PUBLIC - STATE OF ILLINOIS
MY COMMISSION EXPIRES:03/21/15

Appendix "C"



446 Hollandview Trail, PO Box 34047 Aurora, ON L4G 7Z9 Tel: (905) 751-5160 Fax: (866) 447-6960

February 27, 2012

blutip Power Technologies Ltd. 6705 Millcreek Dr. Unit #4 Mississauga, Ontario L5N 5M4

Attention: Chairman of the Board, Chief Executive Officer, President

Dear Sirs:

This letter is to advise you that your rights of exploitation of the ICE Fuel Technologies Ltd. Intellectual Property ("ICE IP") originally transferred to you under the terms and conditions of the "Intellectual Property Purchase and Product Development Agreement" dated November 6th, 2009 and the "Termination Agreement" dated January 6th, 2011 are limited to the use of the ICE IP as defined therein.

Specifically, the rights of exploitation are confined to the "Combined Technology" which is defined as the hydrogen generating system developed by Hy-Drive *combined* with the ICE IP. Mr. Peter Williams, past Chairman of the Board was advised of this position prior to his recent resignation.

Please cease and desist from any exploitation of the ICE IP that is outside of the above rights of use, and contact us should you wish to expand such rights.

Yours truly,

Chief Financial Officer

Ronald J. Bogart

c.c. Shannon Brander

Corporate Secretary - blutip Power Technologies Ltd.

Bob Lipic – President, ICE Fuel Technologies Ltd.



Mr. David.Sieradzki
Duff & Phelps
Receiver in respect of blutip Power Technologies Ltd.

March 2nd 2012

VIA EMAIL

Sir:

regarding Blutip Power Technologies Inc.

We have learned that ICE Fuel technologies Ltd. (Canada), in its contract of last January with BPR, did not have license or ownership to the control technology and had licensed the technology to BPR inappropriately.

As of February 23rd, 2012, Aurora Power Solutions Ltd. ("AURORA") purchased all agreements, products and intellectual property of ICE Fuel technologies (USA), including its distribution and licensing relationship with BPR. There is no agreement in force, to allow BPR to market and sell, the controls technology BPR notionally licensed from ICE Fuel Technologies Ltd. (Canada).

Immediately, all sales programs with the BPR controller are not allowed and any actions must be terminated. You must immediately report to AURORA any and all customer programs, tests and disclosures of any product wherein the controller is utilized. You must immediately notify these customers of your inability to continue standalone controller sales and sales of hydrogen injections systems utilizing the supervisory control technology we own.

You should categorize any and all BPR inventory of standalone controllers, and controllers integrated with your hydrogen injection systems that use digital supervisory-control and report such to AURORA. We trust that these instructions will be followed as outlined immediately.

Yours truly,

Curtiss Wm. Krawetz

AURORA Power Solutions Corporation

Goodmans

Barristers & Solicitors

Bay Adelaide Centre 333 Bay Street, Suite 3400 Toronto, Ontario M5H 2S7

Telephone: 416.979.2211 Facsimile: 416.979.1234 goodmans.ca

Direct Line: 416.597.5165 jwadden@goodmans.ca

March 8, 2012

Our File No.: 12.0185

BY EMAIL

Duff & Phelps 200 King Street West, Suite 1002 Toronto, Ontario M5H 3T4

Attention: David Sieradzki

Dear Mr. Sieradzki:

Re: Sparta Capital Ltd. v. blutip Power Technologies Ltd.

As you know, we are counsel for Sparta Capital Ltd. ("Sparta").

Pursuant to the terms of a Master Sales Agency Agreement dated January 7, 2005 (the "MSA"), Sparta purchased hydrogen generating devices ("HGS Units") from blutip Power Technologies Ltd. ("Blutip") that were designed for installation on transport trucks in order to reduce emissions, fuel consumption and increase the horsepower of internal combustion engines.

Through various purchase orders, Sparta pre-paid Blutip in full for 50 HGS Units for a total of \$547,000. For your ease of reference, attached are invoice #100019 and an agreement with Leeds Transit Sales evidencing these transactions. Two HGS Units were sold by Sparta. However, the remaining 48 HGS Units failed to perform properly and to the standard required. Despite being paid in full, Blutip failed to provide Sparta with properly functioning HGS Units. In satisfaction of their inability to provide functioning HGS Units, Blutip agreed to replace all of Sparta's defective HGS Units and provide Sparta with an additional 9 HGS units as consideration for the delay Sparta had experienced.

To date, Blutip has failed to deliver the HGS Units to Sparta.

We understand that Blutip has 57 HGS Units in its possession which are rightfully Sparta's. This letter is to put you on notice that Sparta reserves all rights to title, rights and interests of 57 HGS Units. Please kindly advise when Sparta can arrange to pick-up these 57 HGS Units.

Goodmans

In the meantime, we trust that no steps will be taken to liquidate these assets without our express written consent.

Yours very truly,

GOODMANS LLP

Jason Wadden

JW/bh

Sparta Capital Inc. cc:

J. Wadden per B. Hef.

Brad Halfin (Goodmans LLP)



446 Hollandview Trail, PO Box 34047 Aurora, ON L4G 7Z9 Tel: (905) 751-5160 Fax: (866) 447-6960

February 27, 2012

blutip Power Technologies Ltd. 6705 Millcreek Dr. Unit #4 Mississauga, Ontario L5N 5M4

Attention: Chairman of the Board, Chief Executive Officer, President

Dear Sirs:

This letter is to advise you that your rights of exploitation of the ICE Fuel Technologies Ltd. Intellectual Property ("ICE IP") originally transferred to you under the terms and conditions of the "Intellectual Property Purchase and Product Development Agreement" dated November 6th, 2009 and the "Termination Agreement" dated January 6th, 2011 are limited to the use of the ICE IP as defined therein.

Specifically, the rights of exploitation are confined to the "Combined Technology" which is defined as the hydrogen generating system developed by Hy-Drive *combined* with the ICE IP. Mr. Peter Williams, past Chairman of the Board was advised of this position prior to his recent resignation.

Please cease and desist from any exploitation of the ICE IP that is outside of the above rights of use, and contact us should you wish to expand such rights.

Yours truly,

Chief Financial Officer

Ronald J. Bogart

c.c. Shannon Brander

Corporate Secretary - blutip Power Technologies Ltd.

Bob Lipic - President, ICE Fuel Technologies Ltd.

From: The Cell Com [mailto:bob@thecell.com] **Sent:** Tuesday, January 17, 2012 11:14 AM **To:** Hy-drive Info; andrew@blutipower.com

Subject: Violation of our NDA and current technology

Andrew,

You are currently violating our NDA and technology IP.

http://www.blutipower.com/products/blutip-controller/

Cease and diciest immediately.

Attached are copies of the NDA's.

Best Regards,

Bob Potchen President **The Cell**, Inc. 407-330-7838 Ph 407-330-6843 FX 120 Keyes Court Sanford, FL 32773 Bob@TheCell.com

This electronic message is intended for the use of the individual or company to whom it is addressed. If the intended recipient is a customer of The Cell, Inc, the information contained in this message is considered confidential, proprietary, privileged, and may contain confidential information and work product. It may also contain trade secrets protected by State and Federal law. If you are not the addressee please do not copy or deliver this message to anyone else. You should immediately delete the message without reading the contents and notify us by return email that the message has been misdirected. We apologize for the inconvenience. Any other use of this information is strictly prohibited. If this email is in response to an inquiry, no relationship is established by this response and no such relationship shall arise absent the execution by both you and The Cell, Inc of a written legal agreement. All information is still considered confidential, proprietary, privileged, and may contain confidential information and/or work product.

Appendix "D"

APPENDIX "D" INTEREST CALCULATIONS

(includes \$20,000 in legal fees)

A) October Note

Principal:

\$2,600,000.00

Interest Paid:

\$800,000.00

Term:

October 21, 2011 to May 3, 2012

OIIII.

195 days

Effective Annual

Rate of Interest:

57.59%

(Interest paid / principal x 365/days of term)

B) December Note

Scenario 1 - Interest paid excludes \$45,000 commitment fee and \$3,000 in legal fees

Principal:

\$800,000.00

Interest Paid:

\$220,932.00

Term:

December 29, 2011 to May 3, 2012

126 days

Effective Annual

Rate of Interest:

80.00%

(Interest paid / principal x 365/days of term)

Scenario 2 - Interest paid includes \$45,000 commitment fee and \$3,000 in legal fees

Principal:

\$800,000.00

Interest Paid:

\$268,932.00

Term:

December 29, 2011 to May 3, 2012

126 days

Effective Annual

Rate of Interest:

97.38%

(Interest paid / principal x 365/days of term)

Scenario 3 - Effective Annual Rate of Interest is 59.9%

Principal:

\$800,000.00

Interest:

\$165,432.00

Term:

December 29, 2011 to May 3, 2012

126 days

Effective Annual

Rate of Interest:

59.90%

(Interest paid / principal x 365/days of term)

Based on the foregoing, the amount of interest that would have been paid if the effective annual interest rate was 59.9% is \$165,432, which is **\$103,500** less than what was pre-paid.

Appendix "E"

DUFF & PHELPS CANADA RESTRUCTURING INC., solely in its capacity as court-appointed receiver of Blutip Power Technologies Ltd. and not in its personal or corporate capacity

- and -

CCM MASTER QUALIFIED FUND, LTD., by its Investment Manager, COGHILL CAPITAL MANAGEMENT, LLC, or its designee

ASSET PURCHASE AGREEMENT

March 9, 2012

TABLE OF CONTENTS

		Page
ARTICLE 1	DEFINITIONS AND PRINCIPLES OF INTERPRETATION	2
1.1	Definitions	2
1.2	Certain Rules of Interpretation	
1.3	Entire Agreement	
1.4	Schedules	7
ARTICLE 2	PURCHASE AND SALE	8
2.1	Purchase and Sale of Purchased Assets	
2.2	Excluded Assets	
2.3	Assumed Liabilities	
2.4	Assignment and Assumption of Contracts	
2.5	Purchase Price	
2.6	Satisfaction of Purchase Price	
2.7	Purchase Price Allocation	
2.8	Employees.	
ARTICLE 3	REPRESENTATIONS AND WARRANTIES OF THE PARTIES	
3.1	Representations and Warranties of the Receiver	11
3.2	Representations and Warranties of the Purchaser	11
ARTICLE 4	PROCEDURES	12
4.1	Stalking Horse and Bidding Procedures Order; Vesting Order	12
4.2	Pre-Closing Cooperation	14
4.3	Acquisition of Assets on "As Is, Where Is" Basis	14
4.4	Title and Risk	15
ARTICLE 5	CONDITIONS PRECEDENT	15
5.1	Conditions Precedent of the Purchaser	15
5.2	Conditions Precedent of the Receiver	16
ARTICLE 6	CLOSING AND DELIVERIES	18
6.1	Closing	18
6.2	Receiver's Deliveries	
6.3	Purchaser's Deliveries	19
ARTICLE 7	TERMINATION	20
7.1	Termination	20
7.2	Effects of Termination	
ARTICLE 8	OTHER COVENANTS OF THE PARTIES; GENERAL	21
8.1	Books and Records	21
8.2	Access of the Receiver to Books and Records	
8.3	Tax Matters	22
8.4	Receiver's Capacity	23

TABLE OF CONTENTS

(continued)

		Page
8.5	Receiver Disclosures	24
8.6	Closing Certificate	24
8.7	Notices	
8.8	Assignment	
8.9	Expenses	
8.10	Time of the Essence	
8.11	Enurement	26
8.12	Amendment	26
8.13	Further Assurances	26
8.14	Execution and Delivery	

SCHEDULE A EXCLUDED ASSETS

SCHEDULE B ASSUMED LIABILITIES

SCHEDULE C PATENTS

SCHEDULE D BIDDING PROCEDURES

SCHEDULE E STALKING HORSE AND BIDDING PROCEDURES ORDER

SCHEDULE F CLAIM

THIS ASSET PURCHASE AGREEMENT is made this day of March, 2012

BETWEEN:

DUFF & PHELPS CANADA RESTRUCTURING INC., solely in its capacity as court-appointed receiver of the assets, undertakings and properties of BLUTIP POWER TECHNOLOGIES LTD. and not in its personal or corporate capacity,

(the "Receiver")

- and -

CCM MASTER QUALIFIED FUND, LTD., a company governed by the Laws of the Cayman Islands, by its Investment Manager, COGHILL CAPITAL MANAGEMENT, LLC, or its designee

(the "Purchaser").

RECITALS:

- A. The Receiver was appointed receiver, without security, of the assets, undertakings and properties of Blutip Power Technologies Ltd. ("Blutip") acquired for, or used in relation to a business carried on by Blutip, including all proceeds thereof (the "Property") pursuant to an order (as may be amended or restated from time to time, the "Appointment Order") of the Ontario Superior Court of Justice (Commercial List) (the "Court") dated February 28, 2012;
- B. The Appointment Order authorizes the Receiver to market any or all of the Property, including advertising and soliciting offers in respect of the Property or any part or parts thereof and to negotiate such terms and conditions of sale as the Receiver in its discretion may deem appropriate;
- C. The Purchaser has agreed to act as a "stalking horse bidder" in connection with the sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Property, meaning that, in the absence of the Receiver's acceptance of a bid for the Purchased Assets (as defined herein) made in accordance with the Bidding Procedures (as defined herein) which is superior to this Agreement (as determined by the Receiver in accordance with the Bidding Procedures), the Purchaser has agreed to purchase the Receiver's and Blutip's right, title and interest, if any, in and to Purchased Assets on the terms and subject to the conditions set forth in this Agreement, in accordance with the Bidding Procedures and subject to obtaining the Vesting Order (as defined herein);
- D. The Receiver will obtain the Stalking Horse and Bidding Procedures Order (as defined

herein) authorizing the Receiver to enter into this Agreement and authorizing the sales process with respect to the Receiver's and Blutip's right, title and interest, if any, in and to Purchased Assets pursuant to the Bidding Procedures.

THEREFORE the Parties agree as follows:

ARTICLE 1 DEFINITIONS AND PRINCIPLES OF INTERPRETATION

1.1 Definitions

Whenever used in this Agreement the following words and terms shall have the meanings set out below:

- "Agreement" means this asset purchase agreement, including all schedules, and all amendments or restatements, as permitted, and references to "Article", "Section" or "Schedule" mean the specified Article or Section of, or Schedule to, this Agreement;
- "Ancillary Agreements" means a bills of sale, assignment and assumption agreement, assignment of intellectual property, and such other agreements, documents, assignments, or instruments of transfer and conveyance reasonably satisfactory in form and substance to the Purchaser and the Receiver; none of which shall contain any representations or warranties of the Receiver except for those provided herein;
- "Appointment Costs" means the Purchaser's costs of seeking and obtaining the Appointment Order, up to and including entry and service of the Appointment Order, being \$47,500, which costs are secured by the Security pursuant to paragraph 30 of the Appointment Order;
- "Appointment Date" means February 28, 2012;
- "Appointment Order" has the meaning given in the Recitals;
- "Assumed Liabilities" has the meaning given in Section 2.3;
- "Auction" has the meaning given in Section 4.1(b);
- "Back-Up Bid" has the meaning given in the Bidding Procedures;
- "Bidding Procedures" has the meaning given in Section 4.1(b);
- "Blutip" has the meaning given in the Recitals;
- "Books and Records" means the books and records of Blutip relating to the Purchased Assets, including financial, corporate, operations and sales books, records, books of account, sales and purchase records, lists of suppliers and customers, business reports, plans and projections and all other documents, surveys, plans, files, records, assessments, correspondence and other data and information, financial or otherwise, including all data, information and databases stored on computer-related or other electronic media;

"Business" means the technology business of Blutip, which business relates to the improvement of engine fuel economy and the reduction of fuel particulates through the use of certain proprietary, patented and patent-pending fuel optimization and emissions control systems and all such other commercial activities incidental and ancillary thereto;

"Business Day" means any day other than a Saturday, Sunday or statutory holiday in the Province of Ontario;

"CCM" means CCM Master Qualified Fund, Ltd.;

"Claims" includes claims, demands, complaints, grievances, actions, applications, suits, causes of action, orders, charges, indictments, prosecutions or other similar processes;

"Closing" means the completion of the transactions described in Section 2.1;

"Closing Certificate" has the meaning given in Section 8.6;

"Closing Date" has the meaning given in Section 6.1(a);

"Closing Time" has the meaning given in Section 6.1(b);

"Consent" means any approval, authorization, consent, order, license, permission, permit (including any environmental permit), qualification, exemption or waiver by any Governmental Authority or other Person;

"Contracts" means contracts, licences, leases, agreements, obligations, promises, undertakings, understandings, arrangements, documents, commitments, entitlements or engagements to which Blutip is a party or by which Blutip is bound or under which Blutip has, or will have, any liability or contingent liability (in each case, whether written or oral, express or implied) relating to the Purchased Assets and/or the Business, as same may be amended and/or restated, and including any and all related quotations, orders, proposals or tenders which remain open for acceptance, warranties and guarantees and documents ancillary thereto;

"Court" has the meaning given in the Recitals;

"Credit Bid Amount" means the aggregate amount of: (i) the Note Credit Bid Amount, (ii) the Appointment Costs, and (iii) the Term Sheet Obligations. For greater certainty, any fees, costs or expenses of CCM (including legal fees) in connection with the Transaction, and the preparation, execution and delivery of this Agreement and all documents and instruments executed pursuant to this Agreement shall form part of the Expense Reimbursement and shall be excluded from the Credit Bid Amount;

"Encumbrances" means liens, hypothecs, charges, security interests, pledges, leases, title retention agreements, mortgages, restrictions on use, development or similar agreements, easements, rights-of-way, title defects, options or adverse claims or encumbrances of any kind or character whatsoever;

"Employees" means any and all (i) employees who are currently employed (including full-time,

part-time and temporary employees) in connection with the Business; and (ii) employees who are currently employed in connection with the Business who are on leaves of absence (including maternity leave, parental leave, disability leave, sickness leave, workers' compensation and other statutory leaves). For greater certainty, the term "Employees" shall not include contractors;

"Excluded Assets" means the assets listed in Schedule A;

"Excluded Contract" means any Contract which is identified in writing by the Purchaser to the Receiver as an excluded Contract prior to the Closing Date in accordance with Section 2.4(a);

"Excluded Employee Obligations" means any obligation to recognize the prior service of the Employees, or to deem such service to be service with the Purchaser for any purpose (including for notice of termination and termination and severance pay) other than as required in accordance with applicable Law;

"Expense Reimbursement" has the meaning given in Section 7.2(c);

"Governmental Authorities" means governments, regulatory authorities, governmental departments, agencies, commissions, bureaus, officials, ministers, Crown corporations, courts, bodies, boards, tribunals or dispute settlement panels or other law or regulation-making organizations or entities: (a) having or purporting to have jurisdiction on behalf of any nation, province, territory, state or other geographic or political subdivision thereof; or (b) exercising, or entitled or purporting to exercise any administrative, executive, judicial, legislative, policy, regulatory or taxing authority or power;

"Intellectual Property" means, with respect to the Receiver or Blutip, all rights in and to (a) patents, patent applications and patent disclosures, including without limitation, the Patents, (b) trademarks, trade names and corporate names and including all goodwill associated therewith, (c) works of authorship, copyrightable works, copyrights, (d) Internet addresses, domain names, websites and web pages, and (e) any and all other intellectual property and proprietary rights;

"Laws" means currently existing applicable statutes, by-laws, rules, regulations, orders, ordinances or judgments, in each case of any Governmental Authority having the force of law;

"Note Credit Bid Amount" means \$3,296,500, which represents the portion of the debt outstanding under the Notes which is secured by the Security and is being credit bid by the Purchaser;

"Notes" means, collectively, (a) the convertible senior secured promissory note issued by Blutip to CCM on October 21, 2011 in the principal amount of \$2,600,000; and (b) the convertible senior secured promissory note issued by Blutip to CCM on December 29, 2011 in the principal amount of \$800,000;

"Parties" means the Receiver and the Purchaser collectively, and "Party" means any one of them;

"Patents" means the patents and patent applications listed in Schedule C;

"Person" means any individual, sole proprietorship, partnership, firm, entity, unincorporated association, unincorporated syndicate, unincorporated organization, trust, body corporate, Governmental Authority, and where the context requires any of the foregoing when they are acting as trustee, executor, administrator or other legal representative;

"Property" has the meaning given in the Recitals;

"Purchaser" has the meaning given in the Recitals;

"Purchase Price" has the meaning given in Section 2.5;

"Purchased Assets" means the tangible and intangible properties, assets, interests, rights, Claims and Contracts of the Receiver and/or Blutip related to the Business, wherever located, as of the Closing Date, including without limitation the following assets, if any, to the extent related to the Business:

- (a) all inventory and accounts receivable;
- (b) all Intellectual Property;
- (c) all furniture, fixtures and equipment;
- (d) all rights under non-disclosure or confidentiality, non-compete or non-solicitation agreements with employees and agents or with third parties;
- (e) any rights, Claims or causes of action for Claims arising out of the operation of the Business;
- (f) all goodwill and other intangibles; and
- (g) all other personal property not contemplated by the foregoing;

which for greater certainty does not include the Excluded Assets;

"Qualified Bids" has the meaning given in the Bidding Procedures;

"Receiver" has the meaning given in the Recitals;

"Receiver's Certificate" has the meaning given in the Appointment Order;

"Sale Hearing" has the meaning given in Section 4.1(b);

"Sale Motion" has the meaning given in Section 4.1(b);

"Security" means, collectively, (i) the General Security Agreement granted by Blutip in favour of CCM dated October 21, 2011; and (ii) the Receiver's Borrowings Charge, as defined in the Appointment Order;

"Stalking Horse and Bidding Procedures Order" has the meaning given in Section 4.1(c);

"Successful Bidder" has the meaning given in the Bidding Procedures;

"Target Closing Date" has the meaning given in Section 7.1(d);

"Tax" and "Taxes" includes any taxes, duties, fees, premiums, assessments, imposts, levies and other charges of any kind whatsoever imposed by any Governmental Authority, including all interest, penalties, fines, additions to tax or other additional amounts imposed by any Governmental Authority in respect thereof, and including those levied on, or measured by, income, gross receipts, profits, capital, transfer, land transfer, sales, goods and services, harmonized sales, use, excise, withholding, business, franchising, property, development, occupancy, payroll, health, social services, education and social security taxes, all surtaxes, all customs duties and import and export taxes, countervail and anti-dumping, all licence, franchise and registration fees and all employment insurance, health insurance and Canada, and other government pension plan premiums or contributions;

"Term Sheet" means the Term Sheet between CCM and the Receiver dated February 27, 2012 pursuant to which CCM committed to provide to the Receiver a credit facility in the amount of \$400,000 on the terms and conditions set out therein, as amended from time to time;

"Term Sheet Obligations" means all outstanding obligations of any kind pursuant to the Term Sheet, as evidenced by the Receiver's Certificates, and secured by the Security as of the Closing Date;

"Termination Date" has the meaning given in Section 7.1(e);

"Transaction" means the purchase and sale of all of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets contemplated by this Agreement;

"Transfer Taxes" has the meaning given in Section 8.3(c); and

"Vesting Order" has the meaning given in Section 4.1(d).

1.2 Certain Rules of Interpretation

In this Agreement:

- (a) **Currency** All references to money amounts are to lawful currency of Canada;
- (b) Governing Law This Agreement is a contract made under and shall be governed by and construed in accordance with the Laws of the Province of Ontario and the federal Laws of Canada applicable in the Province of Ontario;
- (c) **Headings** Headings of Articles and Sections are inserted for convenience of reference only and shall not affect the construction or interpretation of this Agreement;
- (d) **Including** Where the word "including" or "includes" is used in this Agreement, it means "including (or includes) without limitation";

- (e) **No Strict Construction** The language used in this Agreement is the language chosen by the Parties to express their mutual intent, and no rule of strict construction shall be applied against any Party;
- (f) **Number and Gender** Unless the context otherwise requires, words importing the singular include the plural and vice versa and words importing gender include all genders;
- (g) Severability If, in any jurisdiction, any provision of this Agreement or its application to any Party or circumstance is restricted, prohibited or unenforceable, such provision shall, as to such jurisdiction, be ineffective only to the extent of such restriction, prohibition or unenforceability without invalidating the remaining provisions of this Agreement and without affecting the validity or enforceability of such provision in any other jurisdiction or without affecting its application to other Parties or circumstances; and
- (h) **Time Periods** Unless otherwise specified, time periods within or following which any payment is to be made or act is to be done shall be calculated by excluding the day on which the period commences and including the day on which the period ends and by extending the period to the next Business Day following if the last day of the period is not a Business Day.

1.3 Entire Agreement

This Agreement and the agreements and other documents required to be delivered pursuant to this Agreement, constitute the entire agreement between the Parties and set out all the covenants, promises, warranties, representations, conditions, understandings and agreements between the Parties relating to the subject matter of this Agreement and supersede all prior agreements, understandings, negotiations and discussions, whether oral or written. There are no covenants, promises, warranties, representations, conditions, understandings or other agreements, oral or written, express, implied or collateral between the Parties in connection with the subject matter of this Agreement except as specifically set forth in this Agreement and the Purchaser shall acquire the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets on an as is and where is basis. Any cost estimates, projections or other predictions contained or referred to in any other material that has been provided to the Purchaser or any of its affiliates, subsidiaries, agents or representatives are not and shall not be deemed to be representations or warranties of the Receiver or any of its affiliates, subsidiaries, agents, employees or representatives.

1.4 Schedules

The schedules to this Agreement, listed below, are an integral part of this Agreement:

ScheduleDescriptionSchedule A-Excluded AssetsSchedule B-Assumed Liabilities

Schedule	Description
Schedule C	 Patents
Schedule D	 Bidding Procedures
Schedule E	 Stalking Horse and Bidding Procedures Order
Schedule F	 Claim

ARTICLE 2 PURCHASE AND SALE

2.1 Purchase and Sale of Purchased Assets

On the Closing Date, subject to the terms and conditions of this Agreement (which conditions, for greater certainty, include the issuance of the Stalking Horse and Bidding Procedures Order, the determination by the Receiver that this Agreement is the Successful Bid (as defined in and determined in accordance with the Bidding Procedures), and the issuance of the Vesting Order), the Receiver shall transfer, sell, convey, assign and deliver unto the Purchaser, and the Purchaser shall acquire and accept all of Blutip's and the Receiver's right, title and interest, if any, in and to the Purchased Assets, which will be, pursuant to the Vesting Order, free and clear of all Encumbrances.

2.2 Excluded Assets

Notwithstanding any provision of this Agreement to the contrary, the Purchased Assets shall not include any of the Excluded Assets and nothing herein shall be deemed to sell, transfer, assign or convey the Excluded Assets.

2.3 Assumed Liabilities

The Purchaser shall assume as of the Closing Date and shall pay, discharge and perform, as the case may be, from and after the Closing Date, the liabilities and obligations with respect to the Business and/or the Purchased Assets listed on Schedule B (collectively, the "Assumed Liabilities").

The Purchaser shall not assume or be liable for any Encumbrances, the Excluded Employee Obligations, or any other liabilities or obligations of any nature whatsoever, other than the Assumed Liabilities.

2.4 Assignment and Assumption of Contracts

(a) The Purchaser covenants to the Receiver that, no later than ten (10) days prior to the Target Closing Date, the Purchaser shall advise the Receiver in writing as to which Contracts shall be Excluded Contracts. For greater certainty any exclusion of Contracts pursuant to this Section 2.4 shall not affect the Purchase Price.

- (b) Subject to any rights of Consent by counterparties thereto, the terms and conditions of this Section 2.4 and the Vesting Order, the Contracts of Blutip, other than the Excluded Contracts, shall form part of the Purchased Assets assigned and transferred to the Purchaser at Closing, the consideration for which is included in the Purchase Price. The Purchaser will assume and agree to perform and discharge the Assumed Liabilities under the Contracts, other than the Excluded Contracts, pursuant to this Agreement and the applicable Ancillary Agreements.
- (c) At or prior to Closing, the Receiver shall use commercially reasonable efforts to obtain all necessary Consents to assign the Contracts (other than the Excluded Contracts) to the Purchaser. The Purchaser shall be responsible for the Receiver's reasonable routine and non-material out of pocket costs and expenses relating thereto (which, for greater certainty, shall include Transfer Taxes, fees or similar charges). If a counterparty to a Contract requires, as a condition to its Consent to the assignment of such Contract, payment of any monetary default which arose or is related to the period prior to the Appointment Date, such amounts shall be payable by the Purchaser.
- (d) Notwithstanding anything in this Agreement to the contrary, this Agreement shall not constitute an agreement to assign any Contract, to the extent such Contract is not assignable under applicable Law without the consent of any other Person party thereto where the Consent of such Person has not been given or received.
- (e) For greater certainty, if any necessary Consent is required to assign a Contract but not obtained, neither the Receiver nor the Purchaser shall be in breach of this Agreement nor shall the Purchase Price be adjusted or the Closing delayed.

2.5 Purchase Price

The purchase price for the Purchased Assets, exclusive of all applicable Transfer Taxes, shall be the aggregate of the following (the "Purchase Price"):

- (a) the Credit Bid Amount; and
- (b) the Assumed Liabilities.

All applicable Transfer Taxes shall be paid by the Purchaser by certified cheque or bank draft, subject to the terms hereof and the availability of any exemptions or elections under any applicable legislation for such applicable Transfer Taxes.

2.6 Satisfaction of Purchase Price

The Purchaser shall satisfy the Purchase Price on Closing by:

(a) providing a credit to Blutip in the amount of the Note Credit Bid Amount against Blutip's obligations under the Notes;

- (b) providing a credit to Blutip in the amount of the Appointment Costs against Blutip's obligations under paragraph 30 of the Appointment Order;
- (c) providing a credit to the Receiver in the amount of the Term Sheet Obligations against the Receiver's obligations under the Term Sheet as evidenced by the Receiver's Certificates;
- (d) delivering to the Receiver, for and on behalf of Blutip, fully executed releases and waivers with respect to all Appointment Costs, all amounts outstanding under the Notes, all amounts outstanding under the Term Sheet and all amounts outstanding under the Receiver's Certificates (in each case, including any accrued interest thereon and all fees thereunder); and
- (e) the assumption by the Purchaser of the Assumed Liabilities.

2.7 Purchase Price Allocation

No later than five (5) days before the Target Closing Date, the Purchaser shall prepare a written initial allocation of the Purchase Price in respect of each of the Purchased Assets. The Parties, acting reasonably, shall agree, prior to the Closing, on an allocation of the Purchase Price for Tax purposes.

2.8 Employees

- (a) Prior to but conditional on Closing and with effect as of the Closing Date, the Purchaser shall offer continuing employment to all Employees, such offers of employment to be on terms no less favourable as to salary or wages, benefits, hours of work, duties and working conditions than those in effect at Closing. The Receiver will cooperate with the Purchaser in giving notice to the Employees concerning such matters referred to in this Section 2.8(a) as are reasonable under the circumstances.
- (b) As of and following the Closing Date, the Purchaser shall assume and be responsible for all liabilities and obligations (other than the Excluded Employee Obligations), whether accrued after, on or before the Closing Date, with respect to those Employees who accept the Purchaser's offer of employment or who otherwise continue employment with the Purchaser and the Purchaser shall indemnify and save Blutip and the Receiver harmless in respect of all liabilities and obligations assumed by the Purchaser pursuant to this Section 2.8(b), including defending Blutip and the Receiver against any claims regarding such liabilities and obligations and paying all damages and all reasonable costs, expenses and legal fees.

ARTICLE 3 REPRESENTATIONS AND WARRANTIES OF THE PARTIES

3.1 Representations and Warranties of the Receiver

The Receiver hereby represents and warrants to the Purchaser, as of the date hereof and as of the Closing Date, the matters set out below.

- (a) The Receiver has been appointed by the Court as receiver of the Property pursuant to the Appointment Order, a copy of which has been provided to the Purchaser.
- (b) Subject to the issuance of the Vesting Order, the Receiver has all necessary power and authority to enter into this Agreement and to carry out its obligations under this Agreement. This Agreement constitutes a valid and binding obligation of the Receiver enforceable against it in accordance with its terms subject to any limitations imposed by Law.
- (c) Blutip is not a non-resident of Canada for the purposes of the *Income Tax Act* (Canada).
- (d) Blutip is registered for harmonized sales tax purposes under Part IX of the *Excise Tax Act* (Canada) with the following registration number: 89179 6674 RT0001.
- (e) The Receiver has not engaged in any act that has or could result in an Encumbrance affecting any of the Purchased Assets (other than any charge created by the Appointment Order or arising by operation of Law in the normal course of the Business).

3.2 Representations and Warranties of the Purchaser

The Purchaser hereby represents and warrants to the Receiver, as of the date hereof and as of the Closing Date, the matters set out below.

- (a) The Purchaser has been duly incorporated and is validly subsisting under the Laws of the jurisdiction of its incorporation, and has all requisite corporate capacity, power and authority to carry on its business as now conducted by it and to own its properties and assets and is qualified to carry on business under the Laws of the jurisdictions where it carries on a material portion of its business.
- (b) The execution, delivery and performance by the Purchaser of this Agreement:
 - (i) has been duly authorized by all necessary corporate action on the part of the Purchaser;
 - (ii) does not (or would not with the giving of notice, the lapse of time or the happening of any other event or condition) require any consent or approval under, result in a breach or a violation of, or conflict with, any of

the terms or provisions of its constating documents or by-laws or any contracts or instruments to which it is a party or pursuant to which any of its assets or property may be affected; and

- (iii) will not result in the violation of any Law.
- (c) This Agreement has been duly executed and delivered by the Purchaser and constitutes legal, valid and binding obligations of the Purchaser, enforceable against it in accordance with its terms subject only to any limitation under applicable Laws relating to: (i) bankruptcy, winding-up, insolvency, arrangement and other similar Laws of general application affecting the enforcement of creditors' rights; and (ii) the discretion that a court may exercise in the granting of equitable remedies such as specific performance and injunction.
- (d) Except for the Vesting Order, no Consent and no declaration to or filing or registration with any Governmental Authority is required in connection with the execution and delivery by the Purchaser of this Agreement or the performance by the Purchaser of its obligations hereunder.
- (e) There is no action, suit, proceeding or claim that is pending or, to the Purchaser's knowledge, threatened in any court or by or before any Governmental Authority that would adversely affect the Purchaser's ability to perform its obligations under this Agreement on a timely basis, other than as set out in Schedule F.
- (f) The Purchaser or its designee will be a registrant for the purposes of the tax imposed under Part IX of the Excise Tax Act (Canada) on or prior to Closing or the Purchaser will be liable for any Tax consequences of not so being.
- (g) The Purchaser or its designee will not be a non-resident of Canada for purposes of the *Income Tax Act* (Canada) on or prior to Closing or the Purchaser will be liable for any Tax consequences of not so being.
- (h) As at Closing, the Purchaser or its designee will be Canadian, or if not Canadian, will qualify as a WTO investor within the meaning of the *Investment Canada Act* (Canada).

ARTICLE 4 PROCEDURES

4.1 Stalking Horse and Bidding Procedures Order; Vesting Order

(a) The Receiver and the Purchaser acknowledge that (i) this Agreement is subject to Court approval, and (ii) Closing the Transaction is subject to this Agreement being determined by the Receiver to be the "Successful Bid" (as defined in and determined in accordance with the Bidding Procedures), and to the issuance of the Vesting Order.

- (b) On or before March 9, 2012, the Receiver shall file and serve a motion (the "Sale Motion"), which Sale Motion shall seek approval of, among other things: (i) the Receiver's execution of this Agreement as a "stalking horse" asset purchase agreement; (ii) a stalking horse sale process and related bidding procedures in connection with the sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets, substantially in the form of Schedule D hereto (the "Bidding Procedures"); and (iii) the scheduling of an auction and sale hearing as contemplated by the Bidding Procedures (the "Auction" and "Sale Hearing", respectively).
- (c) The Receiver shall use its commercially reasonable efforts to have the Court enter on or before March 15, 2012 an order reasonably acceptable to the Purchaser and substantially in the form of Schedule E regarding the matters set out in Section 4.1(b) above (the "Stalking Horse and Bidding Procedures Order").
- (d) If the Receiver does not receive any Qualified Bids (other than this Agreement), the Receiver shall use its commercially reasonable efforts to promptly file and serve a motion with the Court for an Order (the "Vesting Order") in form and substance satisfactory to the Receiver and the Purchaser, acting reasonably, approving the sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets to the Purchaser pursuant to this Agreement and vesting title to the Purchased Assets in the Purchaser free and clear of all Encumbrances.
- (e) If the Receiver receives one or more Qualified Bids (other than this Agreement), the Receiver shall use its commercially reasonable efforts to conduct the Auction for the Purchased Assets on or before April 20, 2012. Upon completion of the Auction, the Receiver shall use its commercially reasonable efforts, to promptly file and serve a motion with the Court for the Vesting Order, approving the sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets to the Successful Bidder and vesting title to the Purchased Assets in the Successful Bidder free and clear of all Encumbrances.
- (f) If the Purchaser is the Successful Bidder (pursuant to this Agreement or otherwise), the Purchaser shall provide any information and take such actions as may be reasonably requested by the Receiver to assist the Receiver in obtaining the Vesting Order and any other order of the Court reasonably necessary to consummate the Transaction. The Purchaser covenants to take, or cause to be taken, all commercially reasonable actions and to do, or cause to be done, all things necessary or proper, consistent with applicable Law, to consummate and make effective the Transaction as soon as possible following the issuance of the Vesting Order, and, in any case, by the Target Closing Date.
- (g) In the event that the Purchaser is the Successful Bidder (pursuant to this Agreement or otherwise), or the Successful Bidder has not yet been determined, and leave to appeal is sought, an appeal is taken or a stay pending appeal is requested with respect to the Stalking Horse and Bidding Procedures Order or the

Vesting Order, the Receiver shall promptly notify the Purchaser of such leave to appeal, appeal or stay request and shall promptly provide to the Purchaser a copy of the related notice(s) or order(s). The Receiver shall also provide the Purchaser with written notice of any motion or application filed in connection with any leave to appeal or appeal from such orders.

4.2 **Pre-Closing Cooperation**

- (a) Prior to the Closing, upon the terms, and subject to the conditions of this Agreement, each of the Parties shall use its commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, and cooperate with each other in order to do, all things necessary, proper or advisable under applicable Law to consummate the Transaction as soon as practicable, including the preparation and filing of all forms, registrations and notices required to be filed to consummate the Closing, and the taking of such actions as are necessary to obtain any requisite Consent, provided that the Receiver shall not be obligated to make any payment or deliver anything of value to any Person (other than filing with and payment of any application fees to Governmental Authorities, all of which shall be paid or reimbursed by the Purchaser) in order to obtain any Consent.
- (b) Each of the Receiver and the Purchaser shall promptly notify the other of the occurrence, to such Party's knowledge, of any event or condition, or the existence, to such Party's knowledge, of any fact, that would reasonably be expected to result in any of the conditions set forth in Section 5.1 or Section 5.2 not being satisfied.

4.3 Acquisition of Assets on "As Is, Where Is" Basis

The Purchaser hereby acknowledges and agrees as follows:

- (a) the Purchased Assets are being purchased on an "as is, where is" basis at the Closing Date;
- (b) it has conducted or will conduct its own searches and investigations relating to the Purchased Assets;
- (c) it has conducted such inspections of the Purchased Assets as it deemed appropriate, satisfied itself with respect to the Purchased Assets and all matters connected with or related to the Purchased Assets, and has relied entirely upon its own investigations and inspections in entering into this Agreement to acquire the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets without regard to any information made available or provided by the Receiver or its officers, directors, employees or agents;
- (d) it will accept the Purchased Assets in their state, condition and location as at the Closing Time and except as expressly set forth in this Agreement, the Receiver makes no representations, warranties, statements or promises on its own behalf or

on behalf of Blutip in favour of the Purchaser concerning the Purchased Assets, or the Receiver's or Blutip's right, title or interest in or to the Purchased Assets, which the Purchaser acknowledges are being acquired on an as-is where-is basis, or the uses or applications of the Purchased Assets, whether express or implied, statutory or collateral, arising by operation of Law or otherwise, including express or implied warranties of merchantability, fitness for a particular purpose, title, description, quantity, condition or quality, and that any and all conditions and warranties expressed or implied by the *Sale of Goods Act* (Ontario) do not apply to the sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets and are hereby waived by the Purchaser; and

(e) without limiting the generality of foregoing, it acknowledges and accepts that the description of the Purchased Assets and any portion thereof contained in the Schedules hereto or otherwise provided by the Receiver is for the purpose of identification only; and that no representation, warranty or condition has or will be given by the Receiver or any other party concerning completeness or the accuracy of such descriptions or with respect to any data room set up by the Receiver.

4.4 Title and Risk

The Purchased Assets shall remain at the risk of the Receiver, to the extent of its interest therein, until Closing and at the risk of the Purchaser from and after Closing. The Receiver covenants to the Purchaser that, during the period from and including the date hereof through and including the Closing Date or the earlier termination of this Agreement, the Receiver shall use commercially reasonable efforts to conduct the Business in substantially the same manner as conducted as of the date hereof, pay all fees with respect to the Patents, to the extent such funding is available to the Receiver, and make all other filings required by Governmental Authorities to enable the Purchaser to maintain the Patents in good standing from and after Closing.

ARTICLE 5 CONDITIONS PRECEDENT

5.1 Conditions Precedent of the Purchaser

The obligations of the Purchaser to complete the purchase of all of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets under this Agreement shall be subject to the satisfaction of or compliance with, at or before the Closing Time, each of the following conditions precedent (each of which is acknowledged to be inserted for the exclusive benefit of the Purchaser and may be waived by it in whole or in part):

(a) all of the representations and warranties of the Receiver made in or pursuant to this Agreement shall be true and correct at the Closing Time and with the same effect as if made at and as of the Closing Time (except as such representations and warranties may be affected by the occurrence of events or transactions expressly contemplated and permitted by this Agreement) and the Purchaser shall

have received a certificate from a senior officer of the Receiver confirming to his knowledge, without personal liability, the truth and correctness of such representations and warranties;

- (b) the Receiver shall have performed or complied with, in all material respects, all its obligations, covenants and agreements under this Agreement;
- (c) the Receiver shall have executed and delivered, or caused to be executed and delivered, to the Purchaser on or prior to the Closing Date the documents required to complete the Transaction as may reasonably be required by the Purchaser or its solicitors;
- (d) there shall be no order issued by any Governmental Authority delaying, restricting or preventing, and no pending or threatened Claim or judicial or administrative proceeding, or investigation against any Party by any Person, for the purpose of enjoining, delaying, restricting or preventing, the consummation of the Transaction or otherwise claiming that this Agreement or the consummation of such transactions is improper or would give rise to proceedings under any Laws;
- (e) the Receiver shall have determined in accordance with the Bidding Procedures that this Agreement is the Successful Bid (as defined in the Bidding Procedures); and
- (f) the Appointment Order, the Stalking Horse and Bidding Procedures Order and the Vesting Order shall have been issued and entered by the Court and such orders shall not have been stayed, vacated or subject to pending appeal and no order shall have been issued which restrains or prohibits the completion of the Transaction.

If any of the foregoing conditions in this Section 5.1 have not been fulfilled by the Termination Date, the Purchaser may terminate this Agreement by notice to the Receiver. However, the Purchaser may waive compliance with any condition in whole or in part if it sees fit to do so, without prejudice to its rights of termination in the event of non-fulfilment of any other condition, in whole or in part, or to its rights to recover damages, if any, for the breach of any representation, warranty, covenant or condition contained in this Agreement.

For the avoidance of doubt, there shall be no conditions precedent to the Purchaser's obligation to consummate the Transaction, except for those conditions precedent specifically set forth in this Section 5.1.

5.2 Conditions Precedent of the Receiver

The obligations of the Receiver to complete the sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets under this Agreement shall be subject to the satisfaction of or compliance with, at or before the Closing Time, each of the following conditions precedent (each of which is acknowledged to be inserted for the exclusive benefit of the Receiver and may be waived by it in whole or in part):

- (a) all of the representations and warranties of the Purchaser made in or pursuant to this Agreement shall be true and correct at the Closing Time and with the same effect as if made at and as of the Closing Time (except as such representations and warranties may be affected by the occurrence of events or transactions expressly contemplated and permitted by this Agreement) and the Receiver shall have received a certificate from a senior officer of the Purchaser confirming to his knowledge, without personal liability, the truth and correctness of such representations and warranties;
- (b) the Purchaser shall have performed or complied with, in all material respects, all its obligations, covenants and agreements under this Agreement;
- (c) the Purchaser shall have executed and delivered or caused to be executed and delivered to the Receiver on or prior to the Closing Date the documents required to complete the Transaction as may reasonably be required by the Receiver or its solicitors;
- (d) there shall be no order issued by any Governmental Authority delaying, restricting or preventing, and no pending or threatened Claim or judicial or administrative proceeding, or investigation against any Party by any Person, for the purpose of enjoining, delaying, restricting or preventing, the consummation of the Transactions or otherwise claiming that this Agreement or the consummation of such transactions is improper or would give rise to proceedings under any Laws;
- (e) the Receiver shall have determined in accordance with the Bidding Procedures that this Agreement is the Successful Bid (as defined in the Bidding Procedures);
- (f) the Appointment Order, the Stalking Horse and Bidding Procedures Order and the Vesting Order shall have been issued and entered by the Court and such orders shall not have been stayed, vacated or subject to pending appeal and no order shall have been issued which restrains or prohibits the completion of the Transaction;
- (g) if the Purchaser assigns any of its rights or obligations arising under this Agreement to an affiliate of the Purchaser at or before Closing, such affiliate shall have executed and delivered an assignment and assumption agreement (pursuant to which the Purchaser shall remain jointly and severally liable) satisfactory to the Receiver, acting reasonably; and
- (h) the Receiver shall have received sufficient funds to satisfy all amounts secured by the Receiver's Charge (as defined in the Appointment Order) to and including the Closing Date and such additional funds necessary to wind down and complete the receivership, in an amount to be agreed upon by the Purchaser and the Receiver, acting reasonably, from the Purchaser, either (i) by way of funds advanced by the Purchaser in its capacity as "Lender" to the Receiver under the Term Sheet and evidenced by Receiver's Certificates (which, for greater certainty shall be included in the Term Sheet Obligations); or (ii) by way of a cash payment by the

Purchaser to the Receiver on the Closing Date (which shall be considered part of the Purchase Price). Any amounts received by the Receiver pursuant to this Section 5.2(h) that are not necessary to wind down and complete the receivership shall be returned to the Purchaser upon the Receiver's discharge.

If any of the foregoing conditions in this Section 5.2 have not been fulfilled by the Target Closing Date, the Receiver may terminate this Agreement by notice to the Purchaser. However, the Receiver may waive compliance with any condition in whole or in part if it sees fit to do so, without prejudice to its rights of termination in the event of non-fulfilment of any other condition, in whole or in part, or to its rights to recover damages, if any, for the breach of any representation, warranty, covenant or condition contained in this Agreement.

For the avoidance of doubt, there shall be no conditions precedent to the Receiver's obligation to consummate the Transaction, except for those conditions precedent specifically set forth in this Section 5.2.

ARTICLE 6 CLOSING AND DELIVERIES

6.1 Closing

- (a) Unless otherwise agreed by the Parties, Closing shall occur as soon as practicable after the satisfaction or waiver of all conditions set out in Sections 5.1 and 5.2 (such Closing date, the "Closing Date").
- (b) Closing shall take place at 10:00 a.m. (the "Closing Time") on the Closing Date at the offices of the Receiver's solicitors, or such other time and location as the Parties may agree upon in writing. Any tender of documents or money hereunder may be made upon the Receiver or the Purchaser or upon the solicitors acting for the Party on whom tender is desired. All proceedings to be taken and all documents to be executed and delivered by all parties at the Closing shall be deemed to have been taken and executed simultaneously and no proceedings shall be deemed to have been taken nor documents executed or delivered until all have been taken, executed and delivered.

6.2 Receiver's Deliveries

At the Closing,

- (a) the sale, transfer, assignment, conveyance and delivery by the Receiver of its and Blutip's right, title and interest, if any, in and to the Purchased Assets to the Purchaser, free and clear of all Encumbrances, shall be effected by the issued and entered Vesting Order and by execution and delivery by the Receiver of the Ancillary Agreements;
- (b) the Receiver shall deliver, pursuant to the Vesting Order, free and clear title and possession of the Purchased Assets on an "as is, where is" basis in accordance

- with Section 4.3, provided that delivery shall occur in situ wherever such Purchased Assets are located on the Closing Date;
- (c) the Receiver shall deliver a true and complete copy of the Vesting Order and the Closing Certificate;
- (d) the Receiver shall deliver a bring-down certificate executed by the Receiver, in a form satisfactory to the Purchaser, acting reasonably, certifying that all of the representations and warranties of the Receiver hereunder remain true and correct in all material respects as of the Closing;
- (e) Blutip or the Receiver, as applicable, shall make the elections referred to in Section 8.3, to the extent such elections are applicable to the Transaction and available to the Purchaser;
- (f) the Receiver shall deliver an agreement executed by the Receiver, in a form satisfactory to the Purchaser, acting reasonably, assigning the Receiver's and Blutip's right, title and interest in and to the Intellectual Property to the Purchaser; and
- (g) the Receiver shall deliver a document setting out the allocation of the Purchase Price, in form and substance satisfactory to the Purchaser, acting reasonably.

6.3 Purchaser's Deliveries

At the Closing,

- (a) the Purchaser shall deliver the releases and waivers set out in Section 2.6(d) executed by the Purchaser, in a form satisfactory to the Receiver, acting reasonably;
- (b) the Purchaser shall make the cash payment referred to in Section 5.2(h)(ii), if applicable;
- (c) the Purchaser shall deliver the Ancillary Agreements to which it is party, executed by the Purchaser, in a form satisfactory to the Receiver, acting reasonably;
- (d) the Purchaser shall deliver a bring-down certificate executed by the Purchaser, in a form satisfactory to the Receiver, acting reasonably, certifying that all of the representations and warranties of the Purchaser hereunder remain true and correct in all material respects as of the Closing;
- (e) the Purchaser shall make the elections referred to in Section 8.3, to the extent such elections are applicable to the Transaction, or shall make a cash payment to the Receiver in an amount sufficient to satisfy any Transfer Taxes;
- (f) the Purchaser shall deliver a document setting out the allocation of the Purchase Price, in form and substance satisfactory to the Receiver, acting reasonably.

ARTICLE 7 TERMINATION

7.1 Termination

This Agreement may be terminated at any time prior to Closing as follows:

- (a) automatically and without any action or notice by either the Receiver to the Purchaser or the Purchaser to the Receiver, immediately upon the selection by the Receiver of a Successful Bid if this Agreement is neither the Successful Bid nor the Back-Up Bid selected at such time;
- (b) subject to any approvals required from the Court, by mutual written consent of the Receiver and the Purchaser;
- (c) automatically and without any action or notice by either the Receiver to the Purchaser or the Purchaser to the Receiver, immediately upon the issuance of a final and nonappealable order, decree, or ruling or any other action by a Governmental Authority to restrain, enjoin or otherwise prohibit the transfer of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets as contemplated hereby;
- (d) by the Receiver if the Closing has not occurred on or before May 3, 2012 (the "Target Closing Date");
- (e) by either the Receiver or the Purchaser if the Closing has not occurred on or before May 15, 2012 (the "**Termination Date**");
- (f) by the Receiver, if there has been a material violation or breach by the Purchaser of any agreement, covenant, representation or warranty which would prevent the satisfaction of any condition set forth in Section 5.2 by the Target Closing Date and such violation or breach has not been waived by the Receiver or cured by the Target Closing Date, unless the Receiver is in material breach of its obligations under this Agreement; and
- (g) by the Purchaser, if there has been a material violation or breach by the Receiver of any agreement, covenant, representation or warranty which would prevent the satisfaction of any condition set forth in Section 5.1 by the Termination Date and such violation or breach has not been waived by the Purchaser or cured by the Termination Date, unless the Purchaser is in material breach of its obligations under this Agreement.

7.2 Effects of Termination

If this Agreement is terminated pursuant to Section 7.1:

- (a) all further obligations of the Parties under or pursuant to this Agreement shall terminate without further liability of any Party to the other except for the provisions of Section 7.2(c);
- (b) the Purchaser shall return to the Receiver all documents, work papers and other material of the Receiver or Blutip, as the case may be, relating to the Transaction, whether obtained before or after the execution hereof; and
- (c) if this Agreement is terminated pursuant to Section 7.1(a), and the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets are sold pursuant to either the Successful Bid or the Back-Up Bid, the Purchaser shall be entitled to its costs incurred in connection with the development, execution, delivery and approval by the Court of this Agreement (including, without limitation, legal expenses related thereto, and preparing and negotiating this Agreement), up to a maximum of \$75,000, from the proceeds of such sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets (the "Expense Reimbursement").

ARTICLE 8 OTHER COVENANTS OF THE PARTIES; GENERAL

8.1 Books and Records

At Closing, the Receiver shall deliver to the Purchaser, at the Purchaser's sole expense, copies of the Books and Records that relate to the Purchased Assets and that are in the possession of the Receiver or that are reasonably within the Receiver's control, including personal and employment files pertaining to those Employees who accept the Purchaser's offer of employment or who otherwise continue employment with the Purchaser, to the extent permitted by applicable Law. The Purchaser shall honour and observe, in connection with the Transaction, all applicable privacy Laws with respect to the collection, use, transfer and disclosure of personal information about Employees.

8.2 Access of the Receiver to Books and Records

The Receiver, any trustee, trustee in bankruptcy or similar official appointed with respect to Blutip, and each of their representatives shall, for a period of six (6) years from the Closing Date, have access to, and the right to copy, at their expense for *bona fide* business purposes, to the extent necessary or useful in connection with their administration, including the filing of any Tax return or the defence or settlement of any litigation or to comply with any applicable Law and during usual business hours, upon reasonable prior notice to the Purchaser, all books and records relating to the Business, the Purchased Assets and the Assumed Liabilities which are to be transferred and conveyed to the Purchaser pursuant to this Agreement. The Purchaser shall use reasonable efforts to retain and preserve all such Books and Records for such six (6) year period and shall advise the Receiver and any trustee, trustee in bankruptcy or similar official appointed with respect to Blutip (if any) prior to disposing of any such Books and Records so that they may take possession of any such Books and Records, if required.

8.3 Tax Matters

- (a) The Purchaser and the Receiver agree to furnish or cause to be furnished to each other, as promptly as practicable, such information and assistance relating to the Purchased Assets and the Assumed Liabilities as is reasonably necessary for the preparation and filing of any Tax return, claim for refund or other required or optional filings relating to Tax matters, for the preparation for and proof of facts during any Tax audit, for the preparation for any Tax protest, for the prosecution of any suit or other proceedings relating to Tax matters and for the answer to any governmental or regulatory inquiry relating to Tax matters.
- (b) For all Tax purposes, the Purchaser and, to the extent applicable, the Receiver, agree to report the Transaction in a manner consistent with the Purchase Price allocation determined in accordance with Section 2.7, and the Purchaser and the Receiver shall not voluntarily take any action inconsistent therewith in any Tax return, refund claim, litigation or otherwise, unless required by applicable Tax Laws. The Purchaser and the Receiver shall each be responsible for the preparation of their own statements required to be filed under the *Income Tax Act* (Canada) and other similar forms in accordance with applicable Tax Laws.
- (c) All amounts payable by the Purchaser to the Receiver pursuant to this Agreement do not include any federal, provincial, state or local value-added, sale, use, consumption, multi-staged, ad valorum, personal property, customs, excise, stamp, transfer, land transfer, or similar Taxes, duties, or charges, or any recording or filing fees or similar charges (collectively, "Transfer Taxes") and all Transfer Taxes are the responsibility of and for the account of the Purchaser. The Purchaser and the Receiver agree to cooperate to determine the amount of Transfer Taxes payable in connection with the Transaction. If the Receiver is required by applicable Law or by administration thereof to collect any applicable Transfer Taxes from the Purchaser, the Purchaser shall pay such Transfer Taxes to the Receiver on Closing, unless the Purchaser qualifies for an exemption from any such applicable Transfer Taxes, in which case the Receiver shall not collect any such applicable Transfer Taxes from the Purchaser provided the Purchaser, in lieu of payment of such applicable Transfer Taxes to the Receiver, delivers to the Receiver such certificates, elections or other documentation required by applicable Law or the administration thereof to substantiate and affect the exemption claimed by the Purchaser.
- (d) The Purchaser shall indemnify and save the Receiver harmless from and against all claims and demands for payment of the Taxes referenced in this Section 8.3, including penalties and interest thereon and any liability or costs incurred as a result of any failure to pay such Taxes when due.
- (e) If applicable, at the Closing, Blutip and the Purchaser shall jointly execute an election under Section 167(1)(b) of the *Excise Tax Act* (Canada) to have subsection 167(1.1) of the *Excise Tax Act* (Canada) apply in respect of the sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased

Assets under this Agreement. The Purchaser shall file the election in the manner and within the time prescribed by subsection 167(1.1) of the Excise Tax Act (Canada). Notwithstanding anything to the contrary in this Agreement, the Purchaser shall indemnify and hold Blutip and the Receiver, and their respective shareholders, directors, officers and employees, harmless in respect of any harmonized sales tax, goods and services tax, penalties, interest and other amounts which may be assessed against the Receiver and/or Blutip as a result of the Transaction not being eligible for such elections or as a result of the Purchaser's failure to file the election within the prescribed time.

- If applicable, Blutip and the Purchaser shall jointly execute and file an election (f) under subsection 20(24) of the Income Tax Act (Canada) in the manner required by subsection 20(25) of the Income Tax Act (Canada) and under the equivalent or corresponding provisions of any other applicable provincial or territorial statute, in the prescribed manners and within the time period permitted under the Income Tax Act (Canada) and under any other applicable provincial or territorial statute, as to such amount paid by the Receiver on behalf of Blutip to the Purchaser for assuming future obligations. In this regard, the Purchaser and Blutip acknowledge that a portion of the Purchased Assets transferred by the Receiver pursuant to this Agreement and having a value equal to the amount elected under subsection 20(24) of the *Income Tax Act* (Canada) and the equivalent provisions of any applicable provincial or territorial statute, is being transferred by the Receiver on behalf of Blutip as a payment for the assumption of such future obligations by the Purchaser.
- (g) If applicable, Blutip and the Purchaser shall jointly elect in the prescribed form under Section 22 of the *Income Tax Act* (Canada) and the corresponding provisions of any other applicable Tax statute as to the sale of accounts receivable of Blutip and designate in such election an amount equal to the portion of the Purchase Price allocated to such accounts receivable pursuant to the allocation determined in accordance with Section 2.7. This election or these elections, shall be made within the prescribed time for such elections.
- (h) The Purchaser and the Receiver shall also execute and deliver such other Tax elections and forms as they may mutually agree upon.

8.4 Receiver's Capacity

The Purchaser acknowledges and agrees that in all matters pertaining to this Agreement, including in its execution, Duff & Phelps Canada Restructuring Inc. is acting solely in its capacity as receiver of the Property pursuant to the Appointment Order and not in its personal, corporate, or any other capacity and the Receiver and its agents, officers, directors and employees will have no personal or corporate liability under or as a result of this Agreement, or otherwise in connection herewith. Any liability of the Receiver hereunder shall be limited to payment of the Expense Reimbursement from the proceeds of sale of the Receiver's and Blutip's right, title and interest, if any, in and to the Purchased Assets pursuant to the Successful Bid or

Back-Up Bid, as applicable, subject to and in accordance with Section 7.2(c), and in no circumstances will the Receiver be liable for any consequential damages including loss of profit.

8.5 Receiver Disclosures

The Receiver shall be entitled to disclose this Agreement and all information provided by the Purchaser in connection herewith to the Court, to the parties in interest to the proceedings in connection with the receivership of Blutip, and to any parties entitled to "Access" (as defined in and determined in accordance with Section 3 of the Bidding Procedures). The Parties will consult with and be cooperative with each other in respect of any press release or public statement or public communication with respect to this Agreement or Transaction.

8.6 Closing Certificate

The Parties hereby acknowledge and agree that the Receiver shall be entitled to file a certificate, substantially in the form attached to the Vesting Order (the "Closing Certificate"), with the Court upon receiving written confirmation from the Purchaser that all conditions of Closing have been satisfied or waived, and the Receiver shall have no liability to the Purchaser or any other person as a result of filing the Closing Certificate.

8.7 Notices

Any notice, consent or approval required or permitted to be given in connection with this Agreement shall be in writing and shall be sufficiently given if delivered (whether in person, by courier service or other personal method of delivery), or if transmitted by e-mail or facsimile:

(a) in the case of a notice to the Purchaser at:

CCM Master Qualified Fund, Ltd. c/o Coghill Capital Management, LLC One North Wacker Drive, Suite 4350 Chicago, IL 60606

Attention:

James Schuler

Fax:

(312)324-2001

Email:

ischuler@coghillcapital.com

with a copy (which shall not constitute notice) to:

Osler, Hoskin & Harcourt LLP 100 King Street West 1 First Canadian Place, Suite 6100 Toronto, Ontario M5X 1B8

Attention:

Marc Wasserman

Fax:

(416) 862-6666

Email:

mwasserman@osler.com

(b) in the case of a notice to the Receiver at:

Duff & Phelps Canada Restructuring Inc. 200 King Street West Suite 1002 Toronto, Ontario M5H 3T4

Attention:

David Sieradzki

Fax:

(647) 497-9470

Email:

david.sieradzki@duffandphelps.com

with a copy (which shall not constitute notice) to:

Blake Cassels & Graydon LLP 199 Bay Street Suite 4000, Commerce Court West Toronto, ON M5L 1A9

Attention:

Linc Rogers

Fax:

(416) 863-2653

Email:

linc.rogers@blakes.com

Any notice delivered or transmitted to a Party as provided above shall be deemed to have been given and received on the day it is delivered or transmitted, provided that it is delivered or transmitted on a Business Day prior to 5:00 p.m. local time in the place of delivery or receipt. However, if the notice is delivered or transmitted after 5:00 p.m. local time or if such day is not a Business Day then the notice shall be deemed to have been given and received on the next Business Day.

Any Party may, from time to time, change its address by giving notice to the other Party in accordance with the provisions of this Section 8.7.

8.8 Assignment

The Purchaser may at any time assign any of its rights or obligations arising under this Agreement to an affiliate of the Purchaser; provided, however, that in the event of any such assignment, the Purchaser shall be jointly and severally liable for the obligations it assigns and shall not be relieved of any liability or obligation hereunder. Subject to the foregoing, no Party may assign this Agreement or any rights or obligations arising under this Agreement without the prior written consent of the other Party.

8.9 Expenses

Except as set forth in Sections 5.2(h) and 7.2(c), each of the Parties shall pay their respective legal, accounting, and other professional advisory fees, costs and expenses incurred in connection with the Transaction, and the preparation, execution and delivery of this Agreement and all documents and instruments executed pursuant to this Agreement.

8.10 Time of the Essence

Time shall be of the essence in respect of the obligations of the Parties arising prior to Closing under this Agreement.

8.11 Enurement

This Agreement shall enure to the benefit of and be binding upon the Parties and their respective successors (including any successor by reason of amalgamation of any Party) and permitted assigns.

8.12 Amendment

No amendment, supplement, modification or waiver of this Agreement and, unless otherwise specified, no consent or approval by any Party, shall be binding unless executed in writing by the Party to be bound thereby.

8.13 Further Assurances

The Parties shall, with reasonable diligence, do all such things and provide all such reasonable assurances as may be required to consummate the Transaction, and each Party shall provide such further documents or instruments required by any other Party as may be reasonably necessary or desirable to effect the purpose of this Agreement and carry out its provisions, whether before or after the Closing provided that the reasonable costs and expenses of any actions taken after Closing at the request of a Party shall be the responsibility of the requesting Party.

8.14 Execution and Delivery

This Agreement may be executed by the Parties in counterparts and may be executed and delivered by facsimile or other electronic means and all such counterparts and facsimiles (or other electronic deliveries) shall together constitute one and the same agreement.

[Remainder of page intentionally left blank]

IN WITNESS OF WHICH the Parties have executed this Agreement as of the date first written above.

DUFF & PHELPS CANADA RESTRUCTURING INC., solely in its capacity as court-appointed receiver of the assets, undertakings and properties of BLUTIP POWER TECHNOLOGIES LTD. and not in its personal or corporate capacity

Зу:						
	Name:					
	Title:					

COGHILL CAPITAL MANAGEMENT, LLC in its capacity as Investment Manager of CCM MASTER QUALIFIED FUND, LTD.

By:

Name:

Title:

LEGAL_1:22915197.4

SCHEDULE A EXCLUDED ASSETS

The Excluded Assets shall include:

- (a) any asset that otherwise would constitute a Purchased Asset but for the fact that it is conveyed, leased or otherwise disposed of in the ordinary course of business prior to the Closing Date not in violation of this Agreement;
- (b) the corporate books and records of internal corporate proceedings, Tax records, work papers and other records that the Receiver is required by Law to retain, provided that copies of such books and records will be provided to the Purchaser at Closing;
- (c) the rights of the Receiver under this Agreement or any Ancillary Agreement and all cash and non-cash consideration payable or deliverable to the Receiver under this Agreement or any Ancillary Agreement;
- (d) all rights under or arising out of insurance policies not relating to the Business or the Purchased Assets or non-assignable as a matter of law; and
- (e) all Excluded Contracts and rights of the Receiver and Blutip related thereto.

SCHEDULE B ASSUMED LIABILITIES

The Assumed Liabilities shall include:

- (a) Obligations after Receivership all liabilities and obligations of the Receiver, if any, arising from the operation of the Business or the wind down and completion of the receivership, to the extent not paid or discharged at Closing and all obligations secured by Encumbrances ranking in priority to the Security to the extent not paid or discharged on Closing;
- (b) **Obligations under Contracts** all of Blutip's and/or the Receiver's liabilities and obligations arising on or after the Closing Date under the Contracts (other than the Excluded Contracts), and all payments or obligations required to be paid, performed or discharged in connection with the assignment of such Contracts;
- (c) **Obligations after Closing** all liabilities and obligations arising on or after the Closing Date but only to the extent that they relate to or arise out of the operation of the Business on or after the Closing Date or the Purchaser's ownership of the Purchased Assets; and
- (d) *Employee Matters* all liabilities and obligations of the Purchaser pursuant to Section 2.8.

SCHEDULE C PATENTS

Country	Application / Registration Number	Description
CA	2,269,382	Electrode Assembly
US	6,524,453	Electrode Assembly
US	7,651,602	Electrolyzer
US	12/514,962	Gas/Liquid Separator for Hydrogen Generating Apparatus
CA	2,670,138	Gas/Liquid Separator for Hydrogen Generating Apparatus
AU	2006201027	Hydrogen Generating Apparatus
CA	2,538,922	Hydrogen Generating Apparatus
CDN	2,368,508	Hydrogen Generating Apparatus and components therefor
US	7,240,641	Hydrogen Generating Apparatus and components therefor
US	6,817,320	Hydrogen Generating Apparatus and components therefor
US	6,332,434	Hydrogen Generating Apparatus and components therefor
CA	2,670,137	Hydrogen Generating Apparatus with Hydrogen Concentration Sensors
CA	2,534,454	Hydrogen Generating System for Operation with Engine Turbo Condition
US	12/514,571	Hydrogen Generating System for Operation with Engine Turbo Condition
AU	2007209732	Hydrogen Generating System for Operation with Engine Turbo Condition
PCT	PCT/CA2010/000831	Method and System for Improving Fuel Economy and Controlling Engine Emissions
US	7,722,066	Vehicle Operation Assembly

US	12/757,215	Vehicle Operation Assembly
CA	2,673,360	Hydrogen Generating Apparatus and components thereof
CA	2,688,798	Hydrogen Generating Apparatus and Components Therefor
CA	2,287,270	Hydrogen Generating Apparatus and components thereof
CA	2,400,775	Electrolyzer
AU	2005301052	Electrochemical Cell Electrolyte Pumping System
CA	2,586,588	Electrochemical Cell Electrolyte Pumping System

SCHEDULE D BIDDING PROCEDURES

See attached.

Schedule "D" Bidding Procedures

On February 28, 2012, the Ontario Superior Court of Justice, Commercial List (the "Court") granted an order (the "Appointment Order") appointing Duff & Phelps Canada Restructuring Inc., in such capacity, the "Receiver") over all of the assets, undertakings and properties acquired for, or used in relation to a business carried on by Blutip Power Technologies Ltd. (the "Debtor").

On March 9, 2012, the Receiver filed a motion (the "Bidding Procedures Motion") with the Court seeking, among other things, approval of (a) the Receiver's execution of an agreement of purchase and sale of substantially all assets of the Debtor (the "Purchased Assets") between the Debtor and CCM Master Qualified Fund, Ltd. (the "Stalking Horse Bidder") dated March 9, 2012 (the "Stalking Horse Agreement") so as to set a minimum floor price in respect of the sale process for the Purchased Assets; (b) certain protections granted to the Stalking Horse Bidder pursuant to the Stalking Horse Agreement; and (c) certain bidding procedures for the solicitation of offers or proposals for the acquisition of the Purchased Assets (each a "Bid").

On March 15, 2012, the Court entered an order granting the relief requested in the Bidding Procedures Motion, including approval of the following bidding procedures (these "Bidding Procedures"). Accordingly, these Bidding Procedures shall govern the solicitation by the Receiver of Bids that are superior to that contemplated by the Stalking Horse Agreement and the proposed sale of the Purchased Assets.

All denominations are in Canadian Dollars.

1. Assets for Sale

The Receiver is soliciting Bids for the Purchased Assets, in whole but not in part, that are superior to that contemplated by the Stalking Horse Agreement.

2. Bidding Deadline

All Bids must be submitted in accordance with the terms of these Bidding Procedures so that they are actually received by each of the Notice Parties (as defined below) no later than 10:00 a.m. (Toronto time) on April 16, 2012 (the "Bid Deadline"). A Bid received by a Notice Party after the Bid Deadline shall not constitute a Qualified Bid (as defined below).

Written copies of the Bids shall be delivered via e-mail or by personal delivery so that they are received by the Bid Deadline by each of the following parties (collectively, the "**Notice Parties**"):

(a) the Receiver at:

Duff & Phelps Canada Restructuring Inc. 200 King Street West Suite 1002 Toronto, ON M5H 3T4

Attention:

David Sieradzki

E-mail:

david.sieradzki@duffandphelps.com

(b) counsel to the Receiver at:

Blake, Cassels & Graydon LLP 199 Bay Street, Suite 4000, Commerce Court West Toronto ON M5L 1A9

Attention:

Linc Rogers

E-mail:

linc.rogers@blakes.com

3. Access to Due Diligence Materials

Interested bidders that execute a Confidentiality Agreement (as defined below) are eligible to receive a confidential information memorandum describing the opportunity to acquire the Purchased Assets, a copy of the Stalking Horse Agreement, and due diligence access or additional non-public information ("Access") upon receipt by the Receiver of such executed Confidentiality Agreement. Such party's Access shall terminate when the earliest of the following events occur:

- (a) such party does not submit a Bid by the Bid Deadline;
- (b) the Receiver determines that such party does not constitute a Qualified Bidder (as defined below);
- (c) such party does not participate in the Auction (as defined below); or
- (d) at the conclusion of the Auction.

Notwithstanding that a party's Access may continue until the end of the Auction, the Receiver shall not be obligated to furnish any due diligence information after the Bid Deadline.

The Receiver will designate an employee or other representative to coordinate all reasonable requests for due diligence Access or additional non-public information from all parties eligible to receive such access and information in accordance with this Section 3. The Receiver is not responsible for, and will bear no liability with respect to, any information obtained by any party in connection with the sale of the Purchased Assets and will not make any representations as to the accuracy or completeness of such information.

4. **Due Diligence From Bidders**

Each Bidder (as defined below) shall comply with all reasonable requests for additional information by the Receiver regarding such Bidder and its contemplated transaction. Failure by a Bidder to comply with requests for additional information will be a basis for the Receiver to determine that the Bidder is not a Qualified Bidder.

5. Bidding Procedures

The Receiver, with the assistance of its advisors, shall: (a) coordinate the efforts of Bidders in conducting their due diligence investigations, as permitted by the provisions herein; (b) receive Bids from interested parties; (c) determine whether a Bidder is a Qualified Bidder; (c) if the Auction is to take place in accordance with these Bidding Procedures, conduct the Auction in accordance with these Bidding Procedures; and (d) negotiate offers made by Bidders in accordance with these Bidding Procedures. The Receiver shall have the right to adopt such ancillary and procedural rules not otherwise set out herein for these Bidding Procedures (including rules that may depart from those set forth herein) that in its reasonable business judgment will better promote the goals of these Bidding Procedures and facilitate the Auction, if the Auction takes place; provided that the adoption of any rule that materially deviates from these Bidding Procedures shall require an order of the Court. The Receiver shall provide the Stalking Horse Bidder with at least five (5) days notice of any motion to the Court to adopt a rule that materially deviates from these Bidding Procedures.

6. Bid Requirements

To participate in the process detailed by these Bidding Procedures and to otherwise be considered for any purpose hereunder, an interested party (other than the Stalking Horse Bidder) (a "Bidder") must satisfactorily provide the Receiver with each of the following on or before the Bid Deadline:

- (a) <u>Identification of Bidder</u>. Identification of the Bidder and any Principals (as defined below), and the representatives thereof who are authorized to appear and act on their behalf for all purposes regarding the contemplated transaction;
- (b) <u>Corporate Authority</u>. Written evidence of the Bidder's chief executive officer or other appropriate senior executive's approval of the contemplated transaction; and, if the Bidder is an entity specially formed for the purpose of effectuating the contemplated transaction, then the Bidder must furnish written evidence reasonably acceptable to the Receiver of the approval of the contemplated transaction by the equity holder(s) of such Bidder and any guarantor(s) of the Bid (the "**Principals**");
- (c) <u>Confidentiality Agreement</u>. An executed confidentiality agreement in form and substance acceptable to the Receiver and its counsel (a "Confidentiality Agreement");

- (d) <u>Financing Sources</u>. Written evidence of a firm, irrevocable commitment for financing from a creditworthy bank or financial institution that will provide such financing without alteration of condition(s) or delay(s) and that is not contingent, or, other evidence of the Bidder's ability, as determined in the Receiver's reasonable business judgment, to consummate the transaction contemplated by its Bid; and
- (e) <u>Bid</u>. A Bid which satisfies each of the following conditions:
 - (i) Form and Content. The Bid must be in the form of an executed markup of the Stalking Horse Agreement reflecting such Bidder's proposed changes to the Stalking Horse Agreement, and must contain a written and binding commitment by the Bidder to close on the terms and conditions set forth therein as soon as practicable after satisfaction or waiver of all conditions and a covenant to use commercial best efforts to satisfy all conditions by the Target Closing Date (as defined below);
 - (ii) <u>Irrevocable</u>. The Bid must be irrevocable until the earlier of (i) the day on which the Bidder is notified that the Bid is not a Qualified Bid; (ii) the day on which a Successful Bid is selected, if the Bid is neither the Successful Bid nor the Back-Up Bid selected on such day; and (iii) May 15, 2012 (the "**Termination Date**");
 - (iii) <u>Contingencies.</u> The Bid may not be conditional on obtaining financing or any internal approval or on the outcome or review of due diligence. Any other contingencies associated with the Bid may not, in the aggregate, be more burdensome than those set forth in the Stalking Horse Agreement;
 - (iv) No Fees payable to Bidder. The Bid may not request or entitle the Bidder (other than the Stalking Horse Bidder) to any break-up fee, expense reimbursement, termination or similar type of fee or payment;
 - (v) Good-Faith Deposit. The Bid must be accompanied by a cash deposit (the "Good Faith Deposit") equal to fifteen percent (15%) of the total cash purchase price contemplated by such Bid, and the Good Faith Deposit shall be paid to the Receiver, to be held by the Receiver in trust in accordance with these Bidding Procedures; and
 - (vi) Minimum Consideration. The aggregate consideration in a Bid must provide a cash purchase price of at least \$3,919,000. For greater certainty, this figure is an estimate, as of the Target Closing Date, of the aggregate of (a) the Credit Bid Amount (as defined in the Stalking Horse Agreement); (b) the maximum Expense Reimbursement (as defined in the Stalking Horse Agreement) of \$75,000; and (c) \$100,000.

7. Designation as Qualified Bidder

A "Qualified Bidder" is a Bidder that the Receiver, in consultation with its advisors, determines satisfies all of the conditions set out in Section 6 above, and the Bid submitted by such Qualified Bidder is a "Qualified Bid". Upon the receipt from a Bidder of all of the deliveries required under Section 6 above, the Receiver shall notify such Bidder with respect to whether it is a Qualified Bidder as soon as practicable after the Bid Deadline.

As soon as practicable after the Bid Deadline, the Receiver shall provide all Qualified Bidders with a copy of the Opening Bid (as defined below) for the Auction.

For greater certainty, the Stalking Horse Bidder is and is deemed to be a Qualified Bidder for all purposes of these Bidding Procedures (including for purposes of the Auction) and the Stalking Horse Agreement is and is deemed to be a Qualified Bid for all purposes of these Bidding Procedures (including for purposes of the Auction).

8. Auction

Only if the Receiver has determined by April 18, 2012 that there is at least one Qualified Bid (other than the Stalking Horse Agreement) shall the Receiver conduct an auction (the "Auction") to determine the highest and/or best Bid with respect to the Purchased Assets. The Auction shall commence on April 20, 2012, at 10:00 a.m. (Toronto time) at the offices of Blake, Cassels & Graydon LLP, 199 Bay Street, Suite 4000, Commerce Court West, Toronto, Ontario, M5L 1A9.

If the Receiver determines that there is no Qualified Bid (other than the Stalking Horse Agreement), (i) the Auction will not take place; (ii) the Stalking Horse Agreement shall be declared the Successful Bid (as defined below); (iii) the Receiver shall seek approval of, and authority to consummate, the Stalking Horse Agreement and the transactions provided for therein at the Sale Hearing (as defined below); and (iv) the Receiver shall post notice of such facts on its website established in connection with the receivership of the Debtor, at:

http://www.duffandphelps.com/services/investment_banking/Pages/RestructuringCases.aspx

If the Auction does take place, the Auction shall be conducted according to the following procedures:

- (a) <u>Participation At The Auction</u>. The Receiver and its professionals shall direct and preside over the Auction. Only a Qualified Bidder is eligible to participate at the Auction. Only the authorized representatives (including counsel and other advisors) of each of the Qualified Bidders, and the Receiver shall be permitted to attend at the Auction.
- (b) Rounds. Bidding at the Auction shall be conducted in rounds. The highest Qualified Bid at the beginning of the Auction shall constitute the "Opening Bid" for the first round and the highest Overbid (as defined below) at the end of each round shall constitute the "Opening Bid" for the following round. In each round, a Qualified Bidder may submit no more than one Overbid. If at

the end of any round of bidding, a Qualified Bidder (other than the Qualified Bidder that submitted the Opening Bid for such round) did not submit an Overbid, then the Receiver may, in its sole discretion, bar such Qualified Bidder from participating in the next round of bidding at the Auction. Any Qualified Bidder who bids in a round (including the Qualified Bidder that submitted the Opening Bid for such round) shall be entitled to participate in the next round of bidding at the Auction.

- (c) Bid Assessment Criteria. The determination of which Qualified Bid constitutes the Opening Bid in the first round of bidding and the determination of which Overbid constitutes the Opening Bid for each subsequent round of bidding shall take into account any factors the Receiver, with the assistance of its advisors, reasonably deems relevant to the value of the Qualified Bid/Overbid, including, among other things, the following: (i) the amount and nature of the consideration; (ii) the proposed assumption of any liabilities; (iii) the ability of the Qualified Bidder to close the proposed transaction; (iv) the proposed closing date and the likelihood, extent and impact of any potential delays in closing; (v) any purchase-price adjustments; (vi) the net economic effect of any changes from the Stalking Horse Agreement contemplated by the Qualified Bid; (vii) the net after-tax consideration to be received by the Receiver; and (viii) such other considerations as the Receiver deems relevant in its reasonable business judgment.
- (d) Overbids. All Bids made after the Opening Bid shall be Overbids, and shall be made and received on an open, non-confidential basis, and the identity of each Qualified Bidder and all material terms of each Overbid shall be fully disclosed to all other Qualified Bidders participating in the Auction. The Receiver shall maintain a transcript of the Opening Bid and all Overbids made and announced at the Auction, including the Successful Bid and the Back-Up Bid (as defined below).
- (e) <u>Terms of Overbids</u>. An "**Overbid**" is any Bid made at the Auction that satisfies the following conditions:
 - (i) Minimum Consideration

The amount of the cash purchase price consideration of any Overbid shall not be less than the cash purchase price consideration of the Opening Bid of the applicable round of bidding, plus \$100,000 (the "Minimum Overbid Increment") or such lower amount as the Receiver may determine in order to facilitate the Auction; provided, that, without duplication, application of (a) any amounts owing by the Debtor to the Stalking Horse Bidder under the Notes (as defined in the Stalking Horse Agreement), to a maximum of \$3,296,500, (b) any portion of the Appointment Costs (as defined in the Stalking Horse Agreement), and (c) any amounts advanced to the Receiver by the Stalking Horse Bidder under the Term Sheet (as defined in the

Stalking Horse Agreement) and evidenced by the Receiver's Certificates (as defined in the Appointment Order), shall be considered as cash purchase price consideration in connection with any Overbid made by the Stalking Horse Bidder.

The Receiver shall credit the maximum amount of the Expense Reimbursement to each and every Overbid submitted by the Stalking Horse Bidder at the Auction, meaning that if an Overbid made by the Stalking Horse Bidder is the Opening Bid for a round of bidding at the Auction, the cash purchase price consideration of any Overbid made in that round of bidding must exceed the Stalking Horse Bidder's Overbid by the maximum amount of the Expense Reimbursement plus the Minimum Overbid Increment (a total amount of \$175,000).

(ii) Remaining terms are the same as for Qualified Bids

Except as modified herein, an Overbid must comply with the conditions for a Bid set forth in Section 6(e) above, provided, however, that the Bid Deadline shall not apply and Overbids need not be accompanied by additional cash deposits during the Auction.

To the extent not previously provided (which shall be determined by the Receiver), a Qualified Bidder submitting an Overbid must submit, as part of its Overbid, written evidence (in the form of financial disclosure or credit-quality support information or enhancement reasonably acceptable to the Receiver) demonstrating such Qualified Bidder's ability to close the transaction contemplated by its Overbid.

(iii) Announcing Highest Overbids

At the end of each round of bidding, the Receiver, with the assistance of its advisors, shall (i) immediately review each Overbid made in such round, (ii) identify the highest and/or best such Overbid, and (iii) announce the identity of the Qualified Bidder and the terms of such highest and/or best Overbid to all Qualified Bidders entitled to participate in the next round of bidding. Such highest and/or best Overbid shall be the Opening Bid for the next round of the Auction.

(f) Adjournments. The Receiver reserves the right, in its reasonable business judgment, to make one or more adjournments in the Auction to, among other things: (A) facilitate discussions between the Receiver and individual Qualified Bidders; (B) allow individual Qualified Bidders to consider how they wish to proceed; (C) consider and determine the current highest and/or best Overbid at any given time during the Auction; and (D) give Qualified Bidders the opportunity to provide the Receiver with such additional evidence as it may require, in its reasonable business judgment, that the Qualified Bidder has obtained all required internal corporate approvals, has sufficient

internal resources, or has received sufficient non-contingent debt and/or equity funding commitments, to consummate the proposed transaction at the prevailing Overbid amount.

Closing the Auction. If, in any round of bidding, no new Overbid is made, the Auction shall be closed and the Receiver shall, with the assistance of its advisors: (i) declare the last Opening Bid as the successful Bid (the "Successful Bid" and the entity or entities submitting such Successful Bid, the "Successful Bidder"); (ii) immediately review the other Overbids made in the previous round or the Qualified Bids if there were no Overbids made at the Auction, and identify and record the next highest and/or best Overbid or Qualified Bid to the Successful Bid (the "Back-Up Bid" and the entity or entities submitting such Back-Up Bid, the "Back-Up Bidder"); and (iii) advise the Successful Bidder and the Back-Up Bidder of such determinations and all other Qualified Bidders that they are not the Successful Bidder or Back-Up Bidder.

To the extent not already provided, the Successful Bidder and the Back-Up Bidder shall each, within two (2) business days of the conclusion of the Auction, provide the Receiver with an additional cash deposit to increase its original Good Faith Deposit to equal fifteen percent (15%) of the total cash purchase price contemplated by the Successful Bid or Back-Up Bid, as applicable, to be held by the Receiver in trust as such party's "Good Faith Deposit" in accordance with these Bidding Procedures. For greater certainty, if the Stalking Horse Bidder submits an Overbid which is determined to be the Successful Bid or Back-Up Bid, the Stalking Horse Bidder shall, within two (2) business days of the conclusion of the Auction, provide the Receiver with a cash deposit equal to fifteen percent (15%) of the total cash purchase price (excluding any amounts which are credit bid pursuant to Section 13 hereof) contemplated by such Overbid, to be held by the Receiver in trust as the Stalking Horse Bidder's "Good Faith Deposit" in accordance with these Bidding Procedures.

- (h) <u>Consent to Jurisdiction as Condition to Bid.</u> All Qualified Bidders at the Auction shall be deemed to have consented to the jurisdiction of the Court and waived any right to a jury trial in connection with any disputes relating to the Auction, and the construction and enforcement of the Qualified Bidder's transaction document, as applicable.
- (i) Expense Reimbursement. In the event that the Stalking Horse Bidder is neither the Successful Bidder nor the Back-Up Bidder, the Stalking Horse Agreement shall be terminated pursuant to Section 7.1(a) of the Stalking Horse Agreement, and the Expense Reimbursement (in the maximum amount of \$75,000) shall be payable to the Stalking Horse Bidder in accordance with the terms of the Stalking Horse Agreement from the proceeds received upon closing of the Successful Bid or the Back-Up Bid, as applicable. Section

7.2(c) of the Stalking Horse Agreement and the rights and obligations created thereunder shall survive termination of the Stalking Horse Agreement.

9. Sale Hearing

The Receiver shall, within seven (7) days of the conclusion of the Auction, or, if there is no Auction as determined in accordance with Section 8, by April 20, 2012, serve notice of a hearing (the "Sale Hearing") to approve the sale of the Purchased Assets to the Successful Bidder free and clear of all liens and encumbrances, other than those liens and encumbrances expressly to be assumed by the Successful Bidder. The Sale Hearing shall be conducted by the Court as soon as possible thereafter, and in any case, by May 1, 2012, at 330 University Avenue, Toronto, Ontario. At the Sale Hearing, the Receiver may also seek, in its sole discretion, conditional approval of the Back-Up Bid, authorizing the Receiver to close the Back-Up Bid if the Successful Bid is not closed by May 3, 2012 (the "Target Closing Date").

10. Closing the Successful Bid

The Receiver and the Successful Bidder shall take all reasonable steps to complete the sale transaction contemplated by the Successful Bid as soon as possible after the Successful Bid is approved by the Court. The Receiver will be deemed to have accepted the Successful Bid only when the Successful Bid has been approved by the Court. If the transaction contemplated by the Successful Bid has not closed by the Target Closing Date or the Successful Bid is terminated for any reason prior to the Target Closing Date, the Receiver may elect, in its sole discretion to seek to complete the transaction contemplated by the Back-Up Bid, and upon making such election, the Receiver will seek Court approval of the Back-Up Bid (if such approval has not already been obtained) and promptly seek to close the transaction contemplated by the Back-Up Bid after such Court approval. The Back-Up Bid will be deemed to be the Successful Bid and the Receiver will be deemed to have accepted the Back-Up Bid only when the Back-Up Bid has been approved by the Court.

11. "As Is, Where Is"

The sale of the Purchased Assets pursuant to these Bidding Procedures shall be on an "as is, where is" basis and without representations or warranties of any kind, nature, or description by the Debtor or the Receiver, other than such limited market standard representations and warranties as may be expressly set out in the Successful Bid.

12. Return of Good Faith Deposit

- (a) All Good Faith Deposits shall be held in an interest-bearing account until returned to the applicable Bidder or otherwise dealt with in accordance with this Section 12.
- (b) Good Faith Deposits of all Bidders who are determined not to be Qualified Bidders shall be returned to such Bidder two (2) business days after the day on which the Bidder is notified that it is not a Qualified Bidder.

- (c) Good Faith Deposits of all Qualified Bidders other than the Successful Bidder and the Back-Up Bidder (if applicable) shall be returned to such Qualified Bidder two (2) business days after the day on which a Successful Bidder is selected.
- (d) The Good Faith Deposit of the Successful Bidder shall be applied to the purchase price of such transaction at closing. If the Successful Bidder fails to consummate the transaction contemplated by the Successful Bid by the Target Closing Date because of a breach or failure to perform on the part of such Successful Bidder, the Receiver shall be entitled to retain the Good Faith Deposit of the Successful Bidder as part of its damages resulting from the breach or failure to perform by the Successful Bidder. The Good Faith Deposit of the Successful Bidder shall otherwise be returned to the Successful Bidder in accordance with the terms of the Successful Bid.
- (e) If the Back-Up Bid has not been deemed to be the Successful Bid, the Good Faith Deposit of the Back-Up Bidder shall be returned to the Back-Up Bidder as soon as practicable after the earlier of (i) the Termination Date, (ii) the closing of the transaction contemplated by the Successful Bid, and (iii) the date on which the Receiver provides written notice to the Back-Up Bidder that the Receiver will not elect to complete the transaction contemplated by the Back-Up Bid.
- (f) If the Back-Up Bid is deemed to be the Successful Bid, the Good Faith Deposit of the Back-Up Bidder shall be applied to the purchase price of such transaction at closing. If the Back-Up Bidder fails to consummate the transaction contemplated by the Back-Up Bid by the Termination Date because of a breach or failure to perform on the part of such Back-Up Bidder, the Receiver shall be entitled to retain the Good Faith Deposit of the Back-Up Bidder as part of its damages resulting from the breach or failure to perform by the Back-Up Bidder. The Good Faith Deposit of the Back-Up Bidder shall otherwise be returned to the Back-Up Bidder in accordance with the terms of the Back-Up Bid.

13. Credit Bidding

For greater certainty and notwithstanding anything herein to the contrary, any lender under any Receiver's Certificate(s) and the Stalking Horse Bidder, as secured creditor pursuant to the Notes and as the creditor who applied for the Appointment Order, shall be permitted, each in their sole discretion, to credit bid up to the full amount of the debt owed to it under the Receiver's Certificate(s), and/or up to \$3,296,500 owed to it under the Notes, and/or up to the amount of the Appointment Costs, to the extent permitted under applicable law.

SCHEDULE E STALKING HORSE AND BIDDING PROCEDURES ORDER

See attached.

Court File No. CV-12-9622-00CL

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

THE HONOURABLE)	THURSDAY DAY, THE
)	
JUSTICE)	15 TH DAY OF MARCH, 2012

IN THE MATTER OF AN APPLICATION PURSUANT TO SECTION 243(1) OF THE *BANKRUPTCY AND INSOLVENCY ACT*, R.S.C. 1985, c. B-3, AS AMENDED

BETWEEN:

CCM MASTER QUALIFIED FUND, LTD.

Applicant

- and -

BLUTIP POWER TECHNOLOGIES LTD.

Respondent

BIDDING PROCEDURES ORDER

THIS MOTION made by Duff & Phelps Canada Restructuring Inc., in its capacity as court-appointed receiver (in such capacity, the "**Receiver**") of blutip Power Technologies Ltd. (the "**Debtor**") for an Order:

- a) abridging the time for service and validating the service of the Notice of Motion,
 the Motion Record and the First Report of the Receiver dated March 9, 2012 (the "First Report") so that this Motion is properly returnable today and dispensing with further service thereof;
- b) approving the First Report and the activities of the Receiver as described therein;
- c) authorizing and directing the Receiver to execute the agreement of purchase and sale of substantially all assets of the Debtor (the "Purchased Assets") between

the Receiver and CCM Master Qualified Fund, Ltd. (the "Stalking Horse Bidder") dated March 9, 2012 (the "Stalking Horse Agreement") attached hereto as Schedule "A" so as to set a minimum floor price in respect of the sale process for the Purchased Assets;

- d) approving and authorizing the payment of the Expense Reimbursement in the manner provided for in the Stalking Horse Agreement, in conjunction with the Bidding Procedures;
- e) approving the bidding procedures attached as Schedule "D" to the Stalking Horse Agreement (the "Bidding Procedures") to be used in conjunction with the solicitation of offers or proposals for the acquisition of the Purchased Assets;
- f) deeming the Stalking Horse Agreement to be a Qualified Bid (as defined in the Bidding Procedures) and accepted solely for the purposes of the Stalking Horse Bidder's right to participate in the Auction (as defined in the Bidding Procedures); and
- g) ordering that the Receiver's Charge and the Receiver's Borrowings Charge, each as defined in the Order of this Court dated February 28, 2012 (the "Appointment Order"), have the standard priority of such charges;

was heard this day at 330 University Avenue, Toronto, Ontario.

ON READING the material filed, including the Notice of Motion, the First Report, the Stalking Horse Agreement and the Bidding Procedures, and on hearing the submissions of counsel for the Receiver and counsel for the Stalking Horse Bidder and on being advised that the Service List was served with the Motion Record herein;

1. **THIS COURT ORDERS** that all capitalized terms used herein and not otherwise defined shall have the meaning given to them in the Stalking Horse Agreement or the Bidding Procedures.

SERVICE

2. **THIS COURT ORDERS** that the time for service of the Notice of Motion, the Motion Record and the First Report is hereby abridged and validated so that this Motion is properly returnable today and hereby dispenses with further service thereof.

RECEIVER'S ACTIVITIES

3. **THIS COURT ORDERS** that the First Report and the activities of the Receiver as described therein are hereby approved.

STALKING HORSE AGREEMENT

- 4. **THIS COURT ORDERS** that the Receiver is hereby authorized and directed to execute the Stalking Horse Agreement.
- 5. **THIS COURT ORDERS** that the provisions of the Stalking Horse Agreement providing for payment of the Expense Reimbursement are hereby authorized and approved and the Receiver is hereby authorized and directed to pay to the Stalking Horse Bidder, in accordance with Section 7.2(c) of the Stalking Horse Agreement, the Expense Reimbursement from the proceeds of sale of a Successful Bid or a Back-Up Bid in the event that the Stalking Horse Agreement is terminated in accordance with Section 7.1(a) thereof.

BIDDING PROCEDURES

- 6. **THIS COURT ORDERS** that the Bidding Procedures are hereby approved and, if one or more Qualified Bids (other than the Stalking Horse Agreement) are received, the Receiver shall be authorized to conduct the Auction as contemplated therein.
- 7. **THIS COURT ORDERS** that the Stalking Horse Agreement is hereby deemed to be a Qualified Bid and accepted solely for the purposes of the Stalking Horse Bidder's right to participate in the Auction.
- 8. **THIS COURT ORDERS** that nothing herein approves the sale of the Purchased Assets on the terms set out in the Stalking Horse Agreement and that the validity of any sale of the Purchased Assets will be determined on a subsequent motion made to this Court.

PRIORITY OF CHARGES

- 9. **THIS COURT ORDERS** that the Receiver's Charge (as defined in the Appointment Order) shall form a first charge on the Property in priority to all security interests, trusts, liens, charges and encumbrances, statutory or otherwise, in favour of any Person, but subject to sections 14.06(7), 81.4(4), and 81.6(2) of the *Bankruptcy and Insolvency Act*, R.S.C. 1985, c. B-3, as amended (the "**BIA**").
- 10. **THIS COURT ORDERS** that the Receiver's Borrowings Charge (as defined in the Appointment Order) shall form a fixed and specific charge on the whole of the Property in priority to all security interests, trusts, liens, charges and encumbrances, statutory or otherwise, in favour of any Person, but subordinate in priority to the Receiver's Charge and the charges as set out in sections 14.06(7), 81.4(4), and 81.6(2) of the BIA.

GENERAL

- 11. THIS COURT HEREBY REQUESTS the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction in Canada or in the United States to give effect to this Order and to assist the Receiver and its agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Receiver, as an officer of this Court, as may be necessary or desirable to give effect to this Order, to grant representative status to the Receiver in any foreign proceeding, or to assist the Receiver and its agents in carrying out the terms of this Order.
- 12. **THIS COURT ORDERS** that the Receiver be at liberty and is hereby authorized and empowered to apply to any court, tribunal, regulatory or administrative body, wherever located, for the recognition of this Order and for assistance in carrying out the terms of this Order.

SCHEDULE "A"

Stalking Horse Agreement

[Intentionally omitted.]

\equiv
\Box
∞
田
5
Ŏ
\exists
\supseteq
CE
Ξ
\Box
\simeq
ΞŢ
≥
\circ
BLUTIP POWER TECHNOLOGIES
Ţ
\equiv
Ų
Ξ
щ
Ŋ
and
IJ.
COLALIFIED FUND, LTD.

Court File No: CV-12-9622-00CL

Respondent

Applicant

ONTARIO SUPERIOR COURT OF JUSTICE (COMMERCIAL LIST)

Proceeding commenced at Toronto

BIDDING PROCEDURES ORDER

BLAKE, CASSELS & GRAYDON LLP

Suite 4000, Commerce Court West 199 Bay Street, Suite 2800 Toronto, Ontario M5L 1A9

Linc Rogers LSUC No.: 43562N

Tel: (416) 863-4168

Christopher Burr LSUC No.: 55172H Tel: (416) 863-3301

Jenna Willis LSUC No.: 58498U Tel: (416) 863-3348

Fax: (416) 863-2653

Lawyers for the Court-appointed Receiver of Blutip Power Technologies Ltd.

SCHEDULE F

CLAIM

Action commenced by Statement of Claim dated November 17, 2011 of Blutip Financial Corporation against Blutip Power Technologies Ltd. and Coghill Capital Management LLC, Court File No. CV-11-00439702-0000